UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FORM 8-K CURRENT REPORT

Pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934

Date of Report (date of earliest event reported): August 27, 2014

LIBERTY INTERACTIVE CORPORATION

(Exact name of registrant as specified in its charter)

Delaware001-3398284-1288730(State or other jurisdiction of incorporation or organization)(Commission (I.R.S. Employer Identification No.)

12300 Liberty Boulevard Englewood, Colorado 80112

(Address of principal executive offices and zip code)

Registrant's telephone number, including area code: (720) 875-5300

Check the appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligation of the registrant under any of the following provisions (see General Instruction A.2. below):

[] Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)
[] Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)
[] Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))
[] Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))

Item 1.01. Entry into a Material Definitive Agreement Item 2.01. Completion of Acquisition or Disposition of Assets

On August 27, 2014 at 5:00 p.m., New York City time (the "Effective Time"), Liberty Interactive Corporation (the "Company") completed its previously announced spin-off (the "Spin-Off") of its former wholly-owned subsidiary Liberty TripAdvisor Holdings, Inc. ("TripCo").

The Spin-Off was accomplished by the distribution (the "Distribution") by the Company of a dividend of (i) one share of TripCo's Series A common stock for each outstanding share of the Company's Series A LibertyVentures common stock as of 5:00 p.m., New York City time, on August 21, 2014 (such date and time, the "Record Date") and (ii) one share of TripCo's Series B common stock for each outstanding share of the Company's Series B Liberty Ventures common stock as of the Record Date. As a result of the Spin-Off, TripCo is an independent, publicly traded company and its assets and liabilities consist of its 22% economic and 57% voting interest in TripAdvisor, Inc., its 100% ownership interest in BuySeasons, Inc., corporate level cash and cash equivalents of approximately \$50 million and approximately \$400 million indebtedness. Prior to the Spin-Off, TripCo distributed \$350 million in cash to the Company (the source of which was proceeds from TripCo's \$400 million margin loan).

In connection with the Spin-Off, the following agreements were entered into by the Company (the "Spin-Off Agreements"):

- a Reorganization Agreement, dated as of August 15, 2014, between the Company and TripCo, which provides for, among other things, the principal corporate transactions required to effect the Spin-Off, certain conditions to the Spin-Off and provisions governing the relationship between the Company and TripCo with respect to and resulting from the Spin-Off; and
- a Tax Sharing Agreement, dated as of August 27, 2014, between the Company and TripCo, which governs the allocation of taxes, tax benefits, tax items and tax-related losses between the Company and TripCo.

The section of the prospectus forming a part of Amendment No. 4 to TripCo's Registration Statement on Form S-1, filed with the Securities and Exchange Commission on August 11, 2014 (File No. 001-195705), entitled "Certain Relationships and Related Party Transactions—Relationships Between TripCo and Liberty and/or Liberty Media," which describes the material terms of the Spin-Off Agreements, is incorporated herein by reference. These descriptions are qualified in their entirety by reference to the full text of the Spin-Off Agreements, which are filed as Exhibits 2.1 and 10.1, respectively, to this Current Report on Form 8-K.

Item 7.01. Regulation FD Disclosure

On August 28, 2014, the Company announced that Michael George, President and CEO of QVC, Inc., will be presenting at the Goldman Sachs 21st Annual Global Retailing Conference on Thursday, September 4th at 8:50 a.m., E.D.T. at the Plaza Hotel in New York, NY. During his presentation, Mr. George may make observations regarding the Company's financial performance and outlook.

This Item 7.01 of this Current Report on Form 8-K and the press release attached hereto as Exhibit 99.1 are being furnished to the Securities and Exchange Commission under Item 7.01 of Form 8-K in satisfaction of the public disclosure requirements of Regulation FD and shall not be deemed "filed" for any purpose.

Item 8.01. Other Events

On August 27, 2014, the Company issued a press release (the "Press Release") announcing the completion of the Spin-Off. The full text of the Press Release is filed as Exhibit 99.1 to this Current Report on Form 8-K and is incorporated by reference into this Item 8.01.

Item 9.01. Financial Statements and Exhibits

(b) The information required to be filed pursuant to Items 2.01 and 9.01 pursuant to Article 11 of Regulation S-X is filed herewith for the quarter ended March 31, 2014 as Exhibit 99.3. The Company hereby undertakes to amend this Current Report on Form 8-K to file such information for the quarter ended June 30, 2014 as promptly as practicable after TripCo has filed its Quarterly Report on Form 10-Q for the quarter ended June 30, 2014 with the Securities and Exchange Commission.

(d) Exhibits

Exhibit No.	Name	
2.1	Reorganization Agreement, dated as of August 15, 2014, between Liberty	
	Interactive Corporation and Liberty TripAdvisor Holdings, Inc.	
10.1	Tax Sharing Agreement, dated as of August 27, 2014, between Liberty	
	Interactive Corporation and Liberty TripAdvisor Holdings, Inc.	
99.1	Press Release, dated August 27, 2014.	
99.2	Press Release, dated August 28, 2014.	
99.3	Financial Information.	

SIGNATURE

Pursuant to the requirements of the Securities Exchange Act of 1934, as amended, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

Date: September 3, 2014

LIBERTY INTERACTIVE CORPORATION

By:

/s/ Wade Haufschild Name: Wade Haufschild Title: Vice President

EXHIBIT INDEX

Exhibit No.	Name
2.1	Reorganization Agreement, dated as of August 15, 2014, between Liberty
	Interactive Corporation and Liberty TripAdvisor Holdings, Inc.
10.1	Tax Sharing Agreement, dated as of August 27, 2014, between Liberty
	Interactive Corporation and Liberty TripAdvisor Holdings, Inc.
99.1	Press Release, dated August 27, 2014.
99.2	Press Release, dated August 28, 2014.
99.3	Financial Information.
	4

EXECUTION COPY

REORGANIZATION AGREEMENT

between

LIBERTY INTERACTIVE CORPORATION

and

LIBERTY TRIPADVISOR HOLDINGS, INC.

Dated as of August 15, 2014

TABLE OF CONTENTS

ARTICLE I THE RESTRUCTURING 1.1 Restructuring 2.2 Transfer of Spinco Assets and Spinco Businesses; Assumption of Spinco Liabilities 2.3 Third Party Consents and Government Approvals 2.4 Further Actions 3.5 Restructuring Documents 3.6 Qualification as Reorganization 3.7 Qualification as Reorganization 3.7 The Distribution 3.8 ARTICLE II THE DISTRIBUTION 3.1 The Distribution 3.2 Conditions to the Distribution 3.2 Treatment of Outstanding Equity Awards 3.3 Treatment of Outstanding Equity Awards 3.1 Representations and Warranties of the Parties 3.2 No Approvals or Notices Required; No Conflict with Instruments 3.3 No Other Reliance 3.4 RATICLE IV COVENANTS 3.5 Purther Assurances 4.1 Cross-Indemnities 4.2 Financing 4.1 Purther Assurances 4.1 Cross-Information 4.2 Specific Performance 4.3 Further Assurances 4.4 Specific Performance 4.5 Access to Information 4.6 Confidentiality 4.7 Notices Regarding Transferred Assets 4.8 Treatment Of Payments 4.9 Treatment Of Payments 4.1 Closing 5.2 Conditions to Closing 5.3 Deliveries at Closing 5.4 ARTICLE VI CLOSING 6.1 Termination 6.2 Effect of Termination 6.3 Definitions 6.4 RATICLE VI TERMINATION 6.5 Definitions 7.6 Governing Law; Jurisdiction 7.7 Waiver of Jury Trial 7.8 Amendments; Waivers 7.9 Amendments; Waivers 7.10 No Strict Construction; Interpretation 7.11 Conflicts with Tax Sharing Agreement	_		Page
1.2 Transfer of Špinco Assets and Spinco Businesses; Assumption of Spinco Liabilities 2 1.3 Third Party Consents and Government Approvals 2 1.4 Further Actions 3 1.5 Restructuring Documents 3 1.6 Qualification as Reorganization 3 ARTICLE II THE DISTRIBUTION 3 2.1 The Distribution 3 2.2 Conditions to the Distribution 4 2.3 Treatment of Outstanding Equity Awards 5 ARTICLE III REPRESENTATIONS AND WARRANTIES 7 3.1 Representations and Warranties of the Parties 7 3.2 No Approvals or Notices Required; No Conflict with Instruments 7 3.3 No Other Reliance 8 ARTICLE IV COVENANTS 8 4.1 Cross-Indemnities 8 4.2 Financing 11 4.3 Further Assurances 11 4.4 Specific Performance 11 4.5 Access to Information 11 4.6 Confidentiality 13 4.7 Notices Regarding Transferred Assets	ARTI	CLE I THE RESTRUCTURING	2
1.2 Transfer of Spinco Assets and Spinco Businesses; Assumption of Spinco Liabilities 2 1.3 Third Party Consents and Government Approvals 2 1.4 Further Actions 3 1.5 Restructuring Documents 3 1.6 Qualification as Reorganization 3 ARTICLE II THE DISTRIBUTION 3 2.1 The Distribution 3 2.2 Conditions to the Distribution 4 2.3 Treatment of Outstanding Equity Awards 5 ARTICLE III REPRESENTATIONS AND WARRANTIES 7 3.1 Representations and Warranties of the Parties 7 3.2 No Approvals or Notices Required; No Conflict with Instruments 7 3.3 No Other Reliance 8 ARTICLE IV COVENANTS 8 4.1 Cross-Indemnities 8 4.2 Financing 11 4.3 Further Assurances 11 4.4 Specific Performance 11 4.5 Access to Information 11 4.6 Confidentiality	1.1	Restructuring	2
1.3 Third Party Consents and Government Approvals 2 1.4 Further Actions 3 1.5 Restructuring Documents 3 1.6 Qualification as Reorganization 3 ARTICLE II THE DISTRIBUTION 3 2.2 Conditions to the Distribution 4 2.3 Treatment of Outstanding Equity Awards 5 ARTICLE II II REPRESENTATIONS AND WARRANTIES 7 3.1 Representations and Warranties of the Parties 7 3.2 No Approvals or Notices Required; No Conflict with Instruments 7 3.1 Representations and Warranties of the Parties 7 3.1 Representations and Warranties of the Parties 7 3.2 No Approvals or Notices Required; No Conflict with Instruments 7 3.1 Representations and Warranties of the Parties 8 4.1 Cross-Indemnities 8 4.2 Financing 11 4.3 Furnities 8 4.1 Cross-Indemnities 11 4.2 Financing 11	1.2	Transfer of Spinco Assets and Spinco Businesses; Assumption of Spinco Liabilities	2
1.4 Further Actions 3 1.5 Restructuring Documents 3 1.6 Qualification as Reorganization 3 ARTICLE II THE DISTRIBUTION 3 2.1 The Distribution 4 2.2 Conditions to the Distribution 4 2.3 Treatment of Outstanding Equity Awards 5 ARTICLE III REPRESENTATIONS AND WARRANTIES 7 3.1 Representations and Warranties of the Parties 7 3.2 No Approvals or Notices Required; No Conflict with Instruments 7 3.3 No Other Reliance 8 ARTICLE IV COVENANTS 8 4.1 Cross-Indemnities 8 4.2 Financing 11 4.3 Further Assurances 11 4.4 Specific Performance 11 4.5 Access to Information 11 4.6 Confidentiality 13 4.7 Notices Regarding Transferred Assets 14 4.8 Treatment of Payments 14 4.8 Treatment of Payments 14 4.8 Treatment	1.3		
1.5 Restructuring Documents 3 1.6 Qualification as Reorganization 3 ARTICLE II THE DISTRIBUTION 3 2.1 The Distribution 3 2.2 Conditions to the Distribution 4 2.3 Treatment of Outstanding Equity Awards 5 ARTICLE III REPRESENTATIONS AND WARRANTIES 7 3.1 Representations and Warranties of the Parties 7 3.2 No Approvals or Notices Required; No Conflict with Instruments 7 3.3 No Other Reliance 8 ARTICLE IV COVENANTS 8 4.1 Cross-Indemnities 8 4.2 Financing 11 4.3 Further Assurances 11 4.4 Specific Performance 11 4.5 Access to Information 11 4.6 Confidentiality 13 4.7 Notices Regarding Transferred Assets 14 4.8 Treatment Of Payments 14 4.8 Treatment of Payments 14 5.1 Closing 14 5.2 Conditions to Clo	1.4		
1.6 Qualification as Reorganization 3 ARTICLE II THE DISTRIBUTION 3 2.1 The Distribution 4 2.2 Conditions to the Distribution 4 2.3 Treatment of Outstanding Equity Awards 5 ARTICLE III REPRESENTATIONS AND WARRANTIES 7 3.1 Representations and Warranties of the Parties 7 3.2 No Approvals or Notices Required; No Conflict with Instruments 7 3.3 No Other Reliance 8 ARTICLE IV COVENANTS 8 4.1 Cross-Indemnities 8 4.2 Financing 11 4.3 Further Assurances 11 4.4 Specific Performance 11 4.5 Access to Information 11 4.6 Confidentiality 13 4.7 Notices Regarding Transferred Assets 14 4.8 Treatment Of Payments 14 4.7 Notices Regarding Transferred Assets 14 5.1 Closing 14 5.2 Conditions to Closing 14 5.1 Tex			
ARTICLE II THE DISTRIBUTION 3 2.1 The Distribution 3 2.2 Conditions to the Distribution 4 2.3 Treatment of Outstanding Equity Awards 5 ARTICLE III REPRESENT ATIONS AND WARRANTIES 7 3.1 Representations and Warranties of the Parties 7 3.2 No Approvals or Notices Required; No Conflict with Instruments 7 3.3 No Other Reliance 8 ARTICLE IV COVENANTS 8 4.1 Cross-Indemnities 8 4.2 Financing 11 4.3 Further Assurances 11 4.4 Specific Performance 11 4.5 Access to Information 11 4.6 Confidentiality 13 4.7 Notices Regarding Transferred Assets 14 4.8 Treatment Of Payments 14 5.1 Closing 14 5.2 Conditions to Closing 14 5.3 Deliveries at Closing 15 5.ARTICLE VI TERMINATION 16 6.1 Termination 16 6.2 Effect of Termination 16 6.1 Definitions 16 7.1 Definitions 16 8 Binding Effect; Assignment		<u> </u>	
2.1 The Distribution 3 2.2 Conditions to the Distribution 4 2.3 Treatment of Outstanding Equity Awards 5 ARTICLE III REPRESENTATIONS AND WARRANTIES 7 3.1 Representations and Warranties of the Parties 7 3.2 No Approvals or Notices Required; No Conflict with Instruments 7 3.3 No Other Reliance 8 ARTICLE IV COVENANTS 8 4.1 Cross-Indemnities 8 4.2 Financing 11 4.3 Further Assurances 11 4.4 Specific Performance 11 4.5 Access to Information 11 4.6 Confidentiality 13 4.7 Notices Regarding Transferred Assets 14 4.8 Treatment Of Payments 14 4.8 Treatment of Payments 14 4.8 Treatment of Payments 14 5.1 Closing 14 5.2 Conditions to Closing 14 5.1 Closing 15 ARTICLE VI TERMINATION 16			
2.2 Conditions to the Distribution 4 2.3 Treatment of Outstanding Equity Awards 5 ARTICLE III REPRESENTATIONS AND WARRANTIES 7 3.1 Representations and Warranties of the Parties 7 3.2 No Approvals or Notices Required; No Conflict with Instruments 7 3.3 No Other Reliance 8 ARTICLE IV COVENANTS 8 4.1 Cross-Indemnities 8 4.2 Financing 11 4.3 Further Assurances 11 4.4 Specific Performance 11 4.5 Access to Information 11 4.6 Confidentiality 13 4.7 Notices Regarding Transferred Assets 14 4.8 Treatment Of Payments 14 4.7 Notices Regarding Transferred Assets 14 4.8 Treatment of Payments 14 5.1 Closing 14 5.1 Closing 14 5.2 Conditions to Closing 15 6.1			
2.3 Treatment of Outstanding Equity Awards 5 ARTICLE III REPRESENTATIONS AND WARRANTIES 7 3.1 Representations and Warranties of the Parties 7 3.2 No Approvals or Notices Required; No Conflict with Instruments 7 3.3 No Other Reliance 8 ARTICLE IV COVENANTS 8 4.1 Cross-Indemnities 8 4.2 Financing 11 4.3 Further Assurances 11 4.4 Specific Performance 11 4.5 Access to Information 11 4.6 Confidentiality 13 4.7 Notices Regarding Transferred Assets 14 4.8 Treatment Of Payments 14 4.8 Treatment Of Payments 14 5.1 Closing 14 5.2 Conditions to Closing 14 5.1 Treatment Of Payments 15 ARTICLE VI IMSCELLANION 16 6.1 Termination 16 6.2 Effect of Termination	2.2		
ARTICLE III REPRESENTATIONS AND WARRANTIES 7 3.1 Representations and Warranties of the Parties 7 3.2 No Approvals or Notices Required; No Conflict with Instruments 7 3.3 No Other Reliance 8 ARTICLE IV COVENANTS 8 4.1 Cross-Indemnities 8 4.2 Financing 11 4.3 Further Assurances 11 4.4 Specific Performance 11 4.5 Access to Information 11 4.6 Confidentiality 13 4.7 Notices Regarding Transferred Assets 14 4.8 Treatment Of Payments 14 4.8 Treatment Of Payments 14 5.1 Closing 14 5.2 Conditions to Closing 14 5.2 Conditions to Closing 15 5.3 Deliveries at Closing 15 ARTICLE VI TERMINATION 16 6.1 Termination 16 6.2 Effect of Termination 16 7.1 Definitions 16 7.2 No Third-Party Rights 24 7.3 Notices 24 7.5 Binding Effect; Assignment 25 7.6 Governing Law; Jurisdiction 25			
3.1 Representations and Warranties of the Parties 7 3.2 No Approvals or Notices Required; No Conflict with Instruments 7 3.3 No Other Reliance 8 ARTICLE IV COVENANTS 8 4.1 Cross-Indemnities 8 4.2 Financing 11 4.3 Further Assurances 11 4.4 Specific Performance 11 4.5 Access to Information 11 4.6 Confidentiality 13 4.7 Notices Regarding Transferred Assets 14 4.8 Treatment Of Payments 14 4.8 Treatment Of Payments 14 5.1 Closing 14 5.2 Conditions to Closing 14 5.3 Deliveries at Closing 15 ARTICLE VI TERMINATION 16 6.1 Termination 16 6.2 Effect of Termination 16 7.1 Definitions 16 7.2 No Third-Party Rights 24 7.3 Notices 24 7.4			
3.2 No Approvals or Notices Required; No Conflict with Instruments 7 3.3 No Other Reliance 8 ARTICLE IV COVENANTS 8 4.1 Cross-Indemnities 8 4.2 Financing 11 4.3 Further Assurances 11 4.4 Specific Performance 11 4.5 Access to Information 11 4.6 Confidentiality 13 4.7 Notices Regarding Transferred Assets 14 4.8 Treatment Of Payments 14 4.8 Treatment Of Payments 14 4.7 Notices Regarding Transferred Assets 14 4.8 Treatment Of Payments 14 5.1 Closing 14 5.1 Closing 14 5.2 Conditions to Closing 14 5.3 Deliveries at Closing 15 5.4 TERMINATION 16 6.1 Termination 16 6.2 Effect of Termination 16 6.1 Definitions 16 7.1			
3.3 No Other Reliance 8 ARTICLE IV COVENANTS 8 4.1 Cross-Indemnities 8 4.2 Financing 11 4.3 Further Assurances 11 4.4 Specific Performance 11 4.5 Access to Information 11 4.6 Confidentiality 13 4.7 Notices Regarding Transferred Assets 14 4.8 Treatment Of Payments 14 4.8 Treatment Of Payments 14 5.1 Closing 14 5.2 Conditions to Closing 14 5.2 Conditions to Closing 15 ARTICLE VI TERMINATION 16 6.1 Termination 16 6.2 Effect of Termination 16 6.1 Termination 16 6.2 Effect of Termination 16 6.1 Definitions 16 7.1 Definitions 24 7.3 Notices 24 7.5 Binding Effect; Assignment 25			
ARTICLE IV COVENANTS 8 4.1 Cross-Indemnities 8 4.2 Financing 11 4.3 Further Assurances 11 4.4 Specific Performance 11 4.5 Access to Information 11 4.6 Confidentiality 13 4.7 Notices Regarding Transferred Assets 14 4.8 Treatment Of Payments 14 ARTICLE V CLOSING 14 5.1 Closing 14 5.2 Conditions to Closing 14 5.3 Deliveries at Closing 15 ARTICLE VI TERMINATION 16 6.1 Termination 16 6.2 Effect of Termination 16 ARTICLE VII MISCELLANEOUS 16 7.1 Definitions 16 7.2 No Third-Party Rights 24 7.3 Notices 24 7.4 Entire Agreement 24 7.5 Binding Effect; Assignment 25 7.6 Governing Law; Jurisdiction 25 7.7 Waiver of Jury Trial 26 7.8 Severability 26 7.9 Amendments; Waivers 26 7.10 On Dicts with Tax Sharing Agreement 27 </td <td></td> <td></td> <td></td>			
4.1 Cross-Indemnities 8 4.2 Financing 11 4.3 Further Assurances 11 4.4 Specific Performance 11 4.5 Access to Information 11 4.6 Confidentiality 13 4.7 Notices Regarding Transferred Assets 14 4.8 Treatment Of Payments 14 5.1 Closing 14 5.1 Closing 14 5.2 Conditions to Closing 14 5.1 Deliveries at Closing 15 ARTICLE VI TERMINATION 16 6.1 Termination 16 6.2 Effect of Termination 16 ARTICLE VII MISCELLANEOUS 16 7.1 Definitions 16 7.2 No Third-Party Rights 24 7.3 Notices 24 7.4 Entire Agreement 24 7.5 Binding Effect; Assignment 25 7.6 Governing Law; Jurisdiction 25 7.8 Severability 26 <			
4.2 Financing 11 4.3 Further Assurances 11 4.4 Specific Performance 11 4.5 Access to Information 11 4.5 Access to Information 13 4.6 Confidentiality 13 4.7 Notices Regarding Transferred Assets 14 4.8 Treatment Of Payments 14 4.8. Treatment Of Payments 14 5.1 Closing 14 5.1 Closing 14 5.2 Conditions to Closing 14 5.3 Deliveries at Closing 15 ARTICLE VI TERMINATION 16 6.1 Termination 16 6.2 Effect of Termination 16 ARTICLE VII MISCELLANEOUS 16 7.1 Definitions 16 7.1 No Third-Party Rights 24 7.3 Notices 24 7.4 Entire Agreement 25 7.5 Binding Effect; Assignment 25 7.6 Governing Law; Jurisdiction 25			
4.3 Further Assurances 11 4.4 Specific Performance 11 4.5 Access to Information 11 4.6 Confidentiality 13 4.7 Notices Regarding Transferred Assets 14 4.8 Treatment Of Payments 14 ARTICLE V CLOSING 14 5.1 Closing 14 5.2 Conditions to Closing 14 5.3 Deliveries at Closing 14 5.3 Deliveries at Closing 15 ARTICLE VI TERMINATION 16 6.1 Termination 16 6.2 Effect of Termination 16 ARTICLE VII MISCELLANEOUS 16 7.1 Definitions 16 7.2 No Third-Party Rights 24 7.3 Notices 24 7.5 Binding Effect; Assignment 25 7.6 Governing Law; Jurisdiction 25 7.7 Waiver of Jury Trial 26 7.8 Severability 26 7.9 Amendments; Waivers 26 <			
4.4 Specific Performance 11 4.5 Access to Information 11 4.6 Confidentiality 13 4.7 Notices Regarding Transferred Assets 14 4.8 Treatment Of Payments 14 4.8 Toology 14 5.1 Closing 14 5.2 Conditions to Closing 14 5.2 Conditions to Closing 14 5.3 Deliveries at Closing 15 ARTICLE VI TERMINATION 16 6.1 Termination 16 6.2 Effect of Termination 16 ARTICLE VII MISCELLANEOUS 16 7.1 Definitions 16 7.2 No Third-Party Rights 24 7.3 Notices 24 7.4 Entire Agreement 24 7.5 Binding Effect; Assignment 25 7.6 Governing Law; Jurisdiction 25 7.7 Waiver of Jury Trial 26 7.8 Severability 26 7.9 Amendments; Waivers 26 7.10 No Strict Construction; Interpretation 27 7.11 Conflicts with Tax Sharing Agreement 27			
4.5 Access to Information 11 4.6 Confidentiality 13 4.7 Notices Regarding Transferred Assets 14 4.8 Treatment Of Payments 14 ARTICLE V CLOSING 14 5.1 Closing 14 5.2 Conditions to Closing 14 5.2 Conditions to Closing 14 5.3 Deliveries at Closing 15 ARTICLE VI TERMINATION 16 6.1 Termination 16 6.2 Effect of Termination 16 ARTICLE VII MISCELLANEOUS 16 7.1 Definitions 16 7.2 No Third-Party Rights 24 7.3 Notices 24 7.4 Entire Agreement 24 7.5 Binding Effect; Assignment 25 7.6 Governing Law; Jurisdiction 25 7.7 Waiver of Jury Trial 26 7.8 Severability 26 7.9 Amendments; Waivers 26 7.10 No Strict Construction; Interpretation			
4.6 Confidentiality 13 4.7 Notices Regarding Transferred Assets 14 4.8 Treatment Of Payments 14 ARTICLE V CLOSING 14 5.1 Closing 14 5.2 Conditions to Closing 14 5.3 Deliveries at Closing 15 ARTICLE VI TERMINATION 16 6.1 Termination 16 6.2 Effect of Termination 16 ARTICLE VII MISCELLANEOUS 16 7.1 Definitions 16 7.2 No Third-Party Rights 24 7.3 Notices 24 7.4 Entire Agreement 24 7.5 Binding Effect; Assignment 25 7.6 Governing Law; Jurisdiction 25 7.7 Waiver of Jury Trial 26 7.8 Severability 26 7.9 Amendments; Waivers 26 7.10 No Strict Construction; Interpretation 27 7.11 Conflicts with Tax Sharing Agreement 27		1	
4.7 Notices Regarding Transferred Assets 14 4.8 Treatment Of Payments 14 ARTICLE V CLOSING 14 5.1 Closing 14 5.2 Conditions to Closing 14 5.3 Deliveries at Closing 15 ARTICLE VI TERMINATION 16 6.1 Termination 16 6.2 Effect of Termination 16 ARTICLE VII MISCELLANEOUS 16 7.1 Definitions 16 7.2 No Third-Party Rights 24 7.3 Notices 24 7.4 Entire Agreement 24 7.5 Binding Effect; Assignment 25 7.6 Governing Law; Jurisdiction 25 7.7 Waiver of Jury Trial 26 7.8 Severability 26 7.9 Amendments; Waivers 26 7.10 No Strict Construction; Interpretation 27 7.11 Conflicts with Tax Sharing Agreement 27			
4.8 Treatment Of Payments 14 ARTICLE V CLOSING 14 5.1 Closing 14 5.2 Conditions to Closing 14 5.3 Deliveries at Closing 15 ARTICLE VI TERMINATION 16 6.1 Termination 16 6.2 Effect of Termination 16 ARTICLE VII MISCELLANEOUS 16 7.1 Definitions 16 7.2 No Third-Party Rights 24 7.3 Notices 24 7.4 Entire Agreement 24 7.5 Binding Effect; Assignment 25 7.6 Governing Law; Jurisdiction 25 7.7 Waiver of Jury Trial 26 7.8 Severability 26 7.9 Amendments; Waivers 26 7.10 No Strict Construction; Interpretation 27 7.11 Conflicts with Tax Sharing Agreement 27			
ARTICLE V CLOSING 14 5.1 Closing 14 5.2 Conditions to Closing 14 5.3 Deliveries at Closing 15 ARTICLE VI TERMINATION 16 6.1 Termination 16 6.2 Effect of Termination 16 ARTICLE VII MISCELLANEOUS 16 7.1 Definitions 16 7.2 No Third-Party Rights 24 7.3 Notices 24 7.4 Entire Agreement 24 7.5 Binding Effect; Assignment 25 7.6 Governing Law; Jurisdiction 25 7.7 Waiver of Jury Trial 26 7.8 Severability 26 7.9 Amendments; Waivers 26 7.10 No Strict Construction; Interpretation 27 7.11 Conflicts with Tax Sharing Agreement 27			
5.1 Closing 14 5.2 Conditions to Closing 14 5.3 Deliveries at Closing 15 ARTICLE VI TERMINATION 16 6.1 Termination 16 6.2 Effect of Termination 16 ARTICLE VII MISCELLANEOUS 16 7.1 Definitions 16 7.2 No Third-Party Rights 24 7.3 Notices 24 7.4 Entire Agreement 24 7.5 Binding Effect; Assignment 25 7.6 Governing Law; Jurisdiction 25 7.7 Waiver of Jury Trial 26 7.8 Severability 26 7.9 Amendments; Waivers 26 7.10 No Strict Construction; Interpretation 27 7.11 Conflicts with Tax Sharing Agreement 27			
5.2 Conditions to Closing 14 5.3 Deliveries at Closing 15 ARTICLE VI TERMINATION 16 6.1 Termination 16 6.2 Effect of Termination 16 ARTICLE VII MISCELLANEOUS 16 7.1 Definitions 16 7.2 No Third-Party Rights 24 7.3 Notices 24 7.4 Entire Agreement 24 7.5 Binding Effect; Assignment 25 7.6 Governing Law; Jurisdiction 25 7.7 Waiver of Jury Trial 26 7.8 Severability 26 7.9 Amendments; Waivers 26 7.10 No Strict Construction; Interpretation 27 7.11 Conflicts with Tax Sharing Agreement 27			
5.3 Deliveries at Closing 15 ARTICLE VI TERMINATION 16 6.1 Termination 16 6.2 Effect of Termination 16 ARTICLE VII MISCELLANEOUS 16 7.1 Definitions 16 7.2 No Third-Party Rights 24 7.3 Notices 24 7.4 Entire Agreement 24 7.5 Binding Effect; Assignment 25 7.6 Governing Law; Jurisdiction 25 7.7 Waiver of Jury Trial 26 7.8 Severability 26 7.9 Amendments; Waivers 26 7.10 No Strict Construction; Interpretation 27 7.11 Conflicts with Tax Sharing Agreement 27		e e e e e e e e e e e e e e e e e e e	
ARTICLE VI TERMINATION 16 6.1 Termination 16 6.2 Effect of Termination 16 ARTICLE VII MISCELLANEOUS 16 7.1 Definitions 16 7.2 No Third-Party Rights 24 7.3 Notices 24 7.4 Entire Agreement 24 7.5 Binding Effect; Assignment 25 7.6 Governing Law; Jurisdiction 25 7.7 Waiver of Jury Trial 26 7.8 Severability 26 7.9 Amendments; Waivers 26 7.10 No Strict Construction; Interpretation 27 7.11 Conflicts with Tax Sharing Agreement 27			
6.1 Termination 16 6.2 Effect of Termination 16 ARTICLE VII MISCELLANEOUS 16 7.1 Definitions 16 7.2 No Third-Party Rights 24 7.3 Notices 24 7.4 Entire Agreement 24 7.5 Binding Effect; Assignment 25 7.6 Governing Law; Jurisdiction 25 7.7 Waiver of Jury Trial 26 7.8 Severability 26 7.9 Amendments; Waivers 26 7.10 No Strict Construction; Interpretation 27 7.11 Conflicts with Tax Sharing Agreement 27			
6.2 Effect of Termination 16 ARTICLE VII MISCELLANEOUS 16 7.1 Definitions 16 7.2 No Third-Party Rights 24 7.3 Notices 24 7.4 Entire Agreement 24 7.5 Binding Effect; Assignment 25 7.6 Governing Law; Jurisdiction 25 7.7 Waiver of Jury Trial 26 7.8 Severability 26 7.9 Amendments; Waivers 26 7.10 No Strict Construction; Interpretation 27 7.11 Conflicts with Tax Sharing Agreement 27			
ARTICLE VII MISCELLANEOUS 16 7.1 Definitions 16 7.2 No Third-Party Rights 24 7.3 Notices 24 7.4 Entire Agreement 24 7.5 Binding Effect; Assignment 25 7.6 Governing Law; Jurisdiction 25 7.7 Waiver of Jury Trial 26 7.8 Severability 26 7.9 Amendments; Waivers 26 7.10 No Strict Construction; Interpretation 27 7.11 Conflicts with Tax Sharing Agreement 27			
7.1 Definitions 16 7.2 No Third-Party Rights 24 7.3 Notices 24 7.4 Entire Agreement 24 7.5 Binding Effect; Assignment 25 7.6 Governing Law; Jurisdiction 25 7.7 Waiver of Jury Trial 26 7.8 Severability 26 7.9 Amendments; Waivers 26 7.10 No Strict Construction; Interpretation 27 7.11 Conflicts with Tax Sharing Agreement 27			
7.2 No Third-Party Rights 24 7.3 Notices 24 7.4 Entire Agreement 24 7.5 Binding Effect; Assignment 25 7.6 Governing Law; Jurisdiction 25 7.7 Waiver of Jury Trial 26 7.8 Severability 26 7.9 Amendments; Waivers 26 7.10 No Strict Construction; Interpretation 27 7.11 Conflicts with Tax Sharing Agreement 27			
7.3 Notices 24 7.4 Entire Agreement 24 7.5 Binding Effect; Assignment 25 7.6 Governing Law; Jurisdiction 25 7.7 Waiver of Jury Trial 26 7.8 Severability 26 7.9 Amendments; Waivers 26 7.10 No Strict Construction; Interpretation 27 7.11 Conflicts with Tax Sharing Agreement 27			
7.4 Entire Agreement 24 7.5 Binding Effect; Assignment 25 7.6 Governing Law; Jurisdiction 25 7.7 Waiver of Jury Trial 26 7.8 Severability 26 7.9 Amendments; Waivers 26 7.10 No Strict Construction; Interpretation 27 7.11 Conflicts with Tax Sharing Agreement 27		• •	
7.5 Binding Effect; Assignment 25 7.6 Governing Law; Jurisdiction 25 7.7 Waiver of Jury Trial 26 7.8 Severability 26 7.9 Amendments; Waivers 26 7.10 No Strict Construction; Interpretation 27 7.11 Conflicts with Tax Sharing Agreement 27			
7.6Governing Law; Jurisdiction257.7Waiver of Jury Trial267.8Severability267.9Amendments; Waivers267.10No Strict Construction; Interpretation277.11Conflicts with Tax Sharing Agreement27			
7.7Waiver of Jury Trial267.8Severability267.9Amendments; Waivers267.10No Strict Construction; Interpretation277.11Conflicts with Tax Sharing Agreement27			
7.8Severability267.9Amendments; Waivers267.10No Strict Construction; Interpretation277.11Conflicts with Tax Sharing Agreement27			
7.9Amendments; Waivers267.10No Strict Construction; Interpretation277.11Conflicts with Tax Sharing Agreement27		· · · · · · · · · · · · · · · · · · ·	
 7.10 No Strict Construction; Interpretation 7.11 Conflicts with Tax Sharing Agreement 27 			
7.11 Conflicts with Tax Sharing Agreement 27			
7.12 Counterparts	7.12	Counterparts	27

REORGANIZATION AGREEMENT

This **REORGANIZATION AGREEMENT** (together with all Schedules and Exhibits hereto, this "<u>Agreement</u>"), dated as of August 15, 2014, is entered into by and between **LIBERTY INTERACTIVE CORPORATION**, a Delaware corporation ("<u>LIC</u>"), and **LIBERTY TRIPADVISOR HOLDINGS, INC.**, a Delaware corporation ("<u>Spinco</u>"). Certain capitalized terms used herein have the meanings ascribed thereto in Section 7.1.

RECITALS:

WHEREAS, Spinco is and prior to the Spin-Off (as defined below) will be a wholly-owned Subsidiary of LIC;

WHEREAS, the LIC Board has determined that it is appropriate and in the best interests of LIC and its stockholders to reorganize its assets and liabilities by means of the Spin-Off (as defined below) of Spinco, the assets and liabilities of which would consist of all of LIC's approximate 22% ownership interest and 57% voting interest in TripAdvisor, Inc., a Delaware corporation ("<u>TripAdvisor</u>"), LIC's 100% ownership interest in BuySeasons, Inc., as well as anticipated corporate level cash and cash equivalents of \$50 million, and \$400 million in indebtedness at the Effective Time (as defined below);

WHEREAS, the parties desire to effect the transactions contemplated by this Agreement, including the Restructuring (as defined below) and the distribution (the "<u>Distribution</u>"), by means of a dividend, of all of the issued and outstanding shares of common stock of Spinco to the holders of record on the Record Date (as defined below) of LIC's Series A Liberty Ventures common stock, par value \$.01 per share ("<u>LVNTA</u>"), and Series B Liberty Ventures common stock, par value \$.01 per share ("<u>LVNTB</u>" and together with LVNTA, the "<u>Liberty Ventures Common Stock</u>");

WHEREAS, the transactions contemplated by this Agreement, including the Restructuring and the Distribution, have been approved by the LIC Board and, to the extent applicable, the Spinco Board, and are motivated in whole or substantial part by certain substantial corporate business purposes of LIC and Spinco;

WHEREAS, the transactions contemplated by this Agreement, including the Contribution and the Distribution (together, the "Spin-Off") are intended to qualify under, among other provisions, Sections 355 and 368 of the Internal Revenue Code of 1986, as amended (the "Code"), and are expected to accomplish certain corporate business purposes of LIC and Spinco (which corporate business purposes are substantially unrelated to U.S. federal tax matters);

WHEREAS, this Agreement constitutes a "plan of reorganization" within the meaning of Section 368 of the Code and the Treasury Regulations promulgated thereunder; and

WHEREAS, the parties wish to set forth in this Agreement the terms on which, and the conditions subject to which, they intend to implement the measures referred to above and elsewhere herein.

NOW, THEREFORE, in consideration of the foregoing and the mutual representations, warranties, covenants and agreements contained herein, the parties to this Agreement hereby agree as follows:

ARTICLE I THE RESTRUCTURING

1.1 Restructuring

•

- (a) The parties have taken or will take, and have caused or will cause their respective Subsidiaries to take, all actions that are necessary or appropriate to implement and accomplish the transactions contemplated by each of the steps set forth in the Restructuring Plan (collectively, the "Restructuring"); provided, that all of such steps (other than Step 13 in the Restructuring Plan) shall be completed by no later than the Effective Time.
- (b) The Contribution and the Distribution are intended to be part of the same plan of reorganization, even though there may be delays between the completion of certain of the transactions.
- 1.2 Transfer of Spinco Assets and Spinco Businesses; Assumption of Spinco Liabilities

.

On the terms and subject to the conditions of this Agreement, and in furtherance of the Restructuring and the Spin-Off:

- (a) LIC, by no later than immediately before the Effective Time, shall cause all of its (or its Subsidiaries') rights, title and interest in and to all of the Spinco Assets and Spinco Businesses to be contributed, assigned, transferred, conveyed and delivered, directly or indirectly, to Spinco, and Spinco agrees to accept or cause to be accepted all such rights, title and interest in and to all the Spinco Assets and Spinco Businesses. All Spinco Assets are being transferred on an "as is, where is" basis, without any warranty whatsoever on the part of LIC.
- (b) LIC, by no later than immediately before the Effective Time, shall cause all of the Spinco Liabilities to be assigned, directly or indirectly, to Spinco, and Spinco agrees to accept, assume, perform, discharge and fulfill all of the Spinco Liabilities in accordance with their respective terms.
- (c) Upon completion of the transactions contemplated by Sections 1.2(a) and (b) above: (i) Spinco will own, directly or indirectly, the Spinco Businesses and the Spinco Assets and be subject to the Spinco Liabilities; and (ii) LIC will continue to own, directly or indirectly, the LIC Retained Businesses and the LIC Retained Assets and continue to be subject to the LIC Retained Liabilities.

1.3 Third Party Consents and Government Approvals

. To the extent that either the Distribution or any step in the Restructuring Plan requires a consent of any third party or a Governmental Authorization, the parties will use commercially reasonable efforts to obtain each such consent and Governmental Authorization at or prior to the time such consent or Governmental Authorization is required in order to lawfully effect the Distribution and each step in the Restructuring Plan.

1.4 Further Actions

. From and after the Effective Time, upon the reasonable request of a party hereto, each other party hereto will promptly take, or cause its Subsidiaries to promptly take, all commercially reasonable actions necessary or appropriate to fully accomplish the Restructuring and to give effect to the transactions provided for in this Agreement, including each step in the Restructuring Plan, in accordance with the purposes hereof.

1.5 Restructuring Documents

. All documents and instruments used to effect the Restructuring and otherwise to comply with this Agreement shall be in form satisfactory to LIC, Spinco and any additional signatories hereto.

1.6 Qualification as Reorganization

. For U.S. federal income tax purposes, (1) each step of the Restructuring is generally intended to be undertaken in a manner so that no gain or loss is recognized (and no income is taken into account) by LIC, Spinco or their respective Subsidiaries, and (2) the Contribution and the Distribution are intended to qualify as a tax-free reorganization under Sections 368(a) and 355 of the Code.

ARTICLE II THE DISTRIBUTION

2.1 The Distribution

•

- (a) The LIC Board shall have the authority and right: (i) to declare or refrain from declaring the Distribution; (ii) to establish and change the date and time of the record date for the Distribution (the "Record Date"); (iii) to establish and change the date and time at which the Distribution shall be effective (the "Distribution Date"); and (iv) prior to the Distribution Date, to establish and change the procedures for effecting the Distribution; subject, in all cases, to the applicable provisions of the DGCL.
- (b) On the Distribution Date, subject to the conditions to the Distribution set forth in Section 2.2, LIC shall cause to be distributed to the holders of record of Liberty Ventures Common Stock on the Record Date (such holders, the "Liberty Ventures Record Holders"), as a dividend, all the issued and outstanding shares of Spinco Common Stock on the basis of (i) one share of Series A Common Stock, par value \$.01 per share, of Spinco ("Spinco Series A Common Stock") for each share of LVNTA held of record on the Record Date and (ii) one share of Series B Common Stock, par value \$.01 per share, of Spinco ("Spinco Series B Common Stock" and together with the Spinco Series A Common Stock, "Spinco Common Stock") for each share of LVNTB held of record on the Record Date.
- (c) Immediately prior to the Distribution Date and in accordance with the Restructuring Plan, Spinco shall cause the Spinco Charter to be filed with the Delaware Secretary of State, whereupon the issued and then outstanding shares of Spinco Common Stock (all of which shall be owned by LIC), shall automatically be reclassified into: (i) a number of shares of Spinco Series A Common Stock equal to the number of shares of LVNTA outstanding as of the Record Date and (ii) a number of shares of Spinco Series B Common Stock equal to the number of shares of LVNTB outstanding as of the Record Date.
- (d) LIC will take such action, if any, as may be necessary or appropriate under applicable state and foreign securities and "blue sky" laws to permit the Distribution to be effected in compliance, in all material respects, with such laws.

2.2 Conditions to the Distribution

- . The Distribution is subject to the satisfaction of the following conditions:
- (a) the LIC Board, or in the case of determining the Record Date, a committee thereof, shall have taken all necessary corporate action to establish the Record Date and to declare the dividends in order to effect the Distribution in accordance with the LIC Charter and bylaws and the DGCL;
- (b) the private letter ruling received by LIC from the IRS (the "Ruling"), providing to the effect that the Spin-Off will qualify as a tax-free transaction under Sections 355 and 368(a)(1)(D) of the Code, and that for U.S. federal income tax purposes, (i) no gain or loss will be recognized by LIC upon the distribution of Spinco Common Stock in the Spin-Off, and (ii) no gain or loss will be recognized by, and no amount will be included in the income of, holders of Liberty Ventures Common Stock upon the receipt of shares of Spinco Common Stock in the Spin-Off, shall not have been withdrawn, invalidated or modified in an adverse manner,
- (c) LIC shall have received the opinion of Baker Botts L.L.P., in form and substance reasonably acceptable to LIC and which opinion will rely upon the continued validity of the Ruling, providing to the effect that the Spin-Off will qualify as a tax-free transaction under Sections 355 and 368(a)(1)(D) of the Code, and that for U.S. federal income tax purposes, (i) no gain or loss will be recognized by LIC upon the distribution of Spinco Common Stock in the Spin-Off, and (ii) no gain or loss will be recognized by, and no amount will be included in the income of, holders of Liberty Ventures Common Stock upon the receipt of shares of Spinco Common Stock in the Spin-Off;
- (d) the Registration Statement on Form S-1 with respect to the registration under the Securities Act of Spinco Common Stock (the "<u>Registration Statement</u>") shall be effective as of the Distribution Date;
 - (e) the Spinco Common Stock shall have been approved for listing on The NASDAQ Stock Market;
- (f) Spinco and one or more of its Subsidiaries shall have entered into the Margin Loan Agreements, secured by the TripAdvisor Securities, in an aggregate principal amount of \$400 million; and
- (g) any other regulatory or contractual approvals that a committee of the Board determines to obtain shall have been so obtained and be in full force and effect.

The foregoing conditions are for the sole benefit of LIC and shall not in any way limit LIC's right to amend, modify or terminate this Agreement in accordance with Section 6.1. Any of the foregoing conditions set forth in Section 2.2(b), (f) and (g) may be waived by the LIC Board and any determination made by the LIC Board prior to the Distribution concerning the satisfaction or waiver of any condition set forth in this Section 2.2 shall be final and conclusive.

2.3 Treatment of Outstanding Equity Awards

.

(a) Certain current and former employees, non-employee directors and consultants of LIC, the Qualifying Subsidiaries and their respective Subsidiaries have been granted options, stock appreciation rights, and restricted shares in respect of LIC Common Stock pursuant to

various stock incentive plans of LIC administered by the LIC Board (collectively, "Awards"). LIC and Spinco shall use commercially reasonable efforts to take all actions necessary or appropriate so that Awards that are outstanding immediately prior to the Effective Time are adjusted as set forth in this Section 2.3.

(b) Options. As of the Effective Time, and as determined by the LIC Board pursuant to its authority granted under the applicable stock incentive plan of LIC, each holder of a Liberty Ventures Option (whether unvested, partially vested or fully vested) who is or formerly was a director, direct employee or consultant of LIC (each such Liberty Ventures Option, an "Outstanding Liberty Ventures Option"), will receive an option to purchase shares of the corresponding series of Spinco Common Stock (a "Spinco Option") and an adjustment to the Outstanding Liberty Ventures Option (as so adjusted, an "Adjusted Liberty Ventures Option") such that the pre-Spin-Off intrinsic value of the Outstanding Liberty Ventures Option is allocated between the Spinco Option and the Adjusted Liberty Ventures Option.

Except as described herein, all other terms of the Spinco Options and the Adjusted Liberty Ventures Options (including the vesting terms thereof) will, in all material respects, be the same as those of the corresponding Outstanding Liberty Ventures Options; *provided*, that the terms and conditions of exercise of the Spinco Options shall in any event be determined in a manner consistent with Section 409A of the Code.

(c) <u>SARs.</u> As of the Effective Time, and as determined by the L IC Board pursuant to its authority granted under the applicable stock incentive plan of LIC, each holder of a Liberty Ventures SAR (whether unvested, partially vested or fully vested) who is or formerly was a director, direct employee or consultant of LIC (each such Liberty Ventures SAR, an "<u>Outstanding Liberty Ventures SAR</u>"), will receive a stock appreciation right with respect to shares of the corresponding series of Spinco Common Stock (a "<u>Spinco SAR</u>") and an adjustment to the Outstanding Liberty Ventures SAR (as so adjusted, an "<u>Adjusted Liberty Ventures SAR</u>") such that the pre-Spin-Off intrinsic value of the Outstanding Liberty Ventures SAR is allocated between the Spinco SAR and the Adjusted Liberty Ventures SAR.

Except as described herein, all other terms of the Spinco SARs and the Adjusted Liberty Ventures SARs (including the vesting terms thereof) will, in all material respects, be the same as those of the corresponding Outstanding Liberty Ventures SARs; *provided*, that the terms and conditions of exercise of the Spinco SARs shall in any event be determined in a manner consistent with Section 409A of the Code.

- (d) Restricted Stock. Shares of Liberty Ventures Common Stock that are subject to a restricted stock award granted under a stock incentive plan of LIC ("Liberty Ventures Restricted Stock") will participate in the Distribution in the same manner as other outstanding shares of Liberty Ventures Common Stock. Except as described herein, shares of Spinco Common Stock received by such holders of Liberty Ventures Restricted Stock ("Spinco Restricted Stock") will otherwise be subject, in all material respects, to the same terms and conditions (including the vesting terms thereof) as those applicable to such shares of Liberty Ventures Restricted Stock immediately prior to the Effective Time.
- (e) From and after the Effective Time, Spinco Options, Spinco SARs and Spinco Restricted Stock, regardless of by whom held, shall be settled by Spinco pursuant to the terms of the Spinco Transitional Plan. The obligation to deliver (i) shares of Spinco Common Stock upon

the exercise of Spinco Options, (ii) cash or shares of Spinco Common Stock in settlement of Spinco SARs or (iii) shares of Spinco Common Stock upon vesting of shares of Spinco Restricted Stock shall be the sole obligation of Spinco, and LIC shall have no Liability in respect thereof.

- (f) It is intended that the Spinco Transitional Plan be considered, as to any Spinco Option, Spinco SAR or Spinco Restricted Stock that is issued as part of the adjustment provisions of this Section 2.3, to be a successor plan to the stock incentive plan of LIC pursuant to which the corresponding Liberty Ventures Option, Liberty Ventures SAR or Liberty Ventures Restricted Stock was issued, and Spinco shall be deemed to have assumed the obligations under the applicable stock incentive plans of LIC to make the adjustments to the Awards set forth in this Section 2.3.
- (g) With respect to Awards adjusted and any equity awards issued as a result of such adjustments (collectively, "Post Spin Awards"), in each case, pursuant to this Section 2.3, service after the Effective Time as an employee or non-employee director of, or consultant to, LIC, Spinco, any Qualifying Subsidiary or any of their respective Subsidiaries shall be treated as service to LIC and Spinco and their respective Subsidiaries for all purposes under such Post Spin Awards following the Effective Time.
- (h) Neither the Effective Time nor any other transaction contemplated by the Restructuring Plan or this Agreement shall be considered a termination of employment for any employee of LIC, Spinco or any of their respective Subsidiaries for purposes of any Post Spin Award.
- (i) Spinco agrees that, on and after the Effective Date, it shall use its reasonable efforts to cause to be effective under the Securities Act, on a continuous basis, a registration statement on Form S-8 with respect to shares of Spinco Common Stock issuable upon exercise of Spinco Options or settlement of Spinco SARs.

ARTICLE III REPRESENTATIONS AND WARRANTIES

- 3.1 Representations and Warranties of the Parties
- . Each party hereto represents and warrants to the other as follows:
- (a) Organization and Qualification. Such party is a corporation duly organized, validly existing and in good standing under the laws of the state of Delaware, has all requisite corporate power and authority to own, use, lease or operate its properties and assets, and to conduct the business heretofore conducted by it, and is duly qualified to do business and is in good standing in each jurisdiction in which the properties owned, used, leased or operated by it or the nature of the business conducted by it requires such qualification, except in such jurisdictions where the failure to be so qualified and in good standing would not have a material adverse effect on its business, financial condition or results of operations or its ability to perform its obligations under this Agreement.
- (b) <u>Authorization and Validity of Agreement</u>. Such party has all requisite power and authority to execute, deliver and perform its obligations under this Agreement, the agreements and instruments to which it is to be a party required to effect the Restructuring (the "<u>Restructuring Agreements</u>") and the agreements to be delivered by it at the Closing pursuant to

Section 5.3 (the "Other Agreements"). The execution, delivery and performance by such party of this Agreement, the Restructuring Agreements and the Other Agreements and the consummation by it of the transactions contemplated hereby and thereby have been duly and validly authorized by the board of directors, managing members or analogous governing body of such party and, to the extent required by law, its stockholders or members, and no other corporate or other action on its part is necessary to authorize the execution and delivery by such party of this Agreement, the Restructuring Agreements and the Other Agreements, the performance by it of its obligations hereunder and thereunder and the consummation by it of the transactions contemplated hereby and thereby. This Agreement has been, and each of the Restructuring Agreements and each of the Other Agreements, when executed and delivered, will be, duly executed and delivered by such party and each is, or will be, a valid and binding obligation of such party, enforceable in accordance with its terms.

3.2 No Approvals or Notices Required; No Conflict with Instruments

. The execution, delivery and performance by such party of this Agreement, the Restructuring Agreements and the Other Agreements, and the consummation of the transactions contemplated hereby and thereby, do not and will not conflict with or result in a breach or violation of any of the terms or provisions of, constitute a default under, or result in the creation of any lien, charge or encumbrance upon any of its assets pursuant to the terms of, the charter or bylaws (or similar formation or governance instruments) of such party, any indenture, mortgage, deed of trust, loan agreement or other agreement or instrument to which it is a party or by which it or any of its assets are bound, or any law, rule, regulation, judgment, order or decree of any court or governmental authority having jurisdiction over it or its properties.

3.3 No Other Reliance

. In determining to enter into this Agreement, the Restructuring Agreements and the Other Agreements, and to consummate the transactions contemplated hereby and thereby, such party has not relied on any representation, warranty, promise or agreement other than those expressly contained herein or therein, and no other representation, warranty, promise or agreement has been made or will be implied. Except as otherwise expressly set forth herein or in the Restructuring Agreements or the Other Agreements, all Spinco Assets and Spinco Businesses are being transferred on an "as is, where is" basis, at the risk of the transferee, without any warranty whatsoever on the part of the transferor and from and after the Effective Time.

ARTICLE IV COVENANTS

4.1 Cross-Indemnities

.

- (a) Spinco hereby covenants and agrees, on the terms and subject to the limitations set forth in this Article IV, from and after the Closing, to indemnify and hold harmless LIC, its Subsidiaries and their respective current and former directors, officers and employees, and each of the heirs, executors, trustees, administrators, successors and assigns of any of the foregoing (the "<u>LIC Indemnified Parties</u>") from and against any Losses incurred by the LIC Indemnified Parties (in their capacities as such) to the extent arising out of or resulting from any of the following:
 - (i) the conduct of the Spinco Businesses (whether before or after the Closing);

- (ii) the Spinco Assets;
- (iii) the Spinco Liabilities (whether incurred before or after the Closing); or
- (iv) any breach of, or failure to perform or comply with, any covenant, undertaking or obligation of Spinco or any of its Subsidiaries under this Agreement, any Restructuring Agreement or any Other Agreement.
- (b) LIC hereby covenants and agrees, on the terms and subject to the limitations set forth in this Article IV, from and after the Closing, to indemnify and hold harmless Spinco, its Subsidiaries and their respective current and former directors, officers and employees, and each of the heirs, executors, trustees, administrators, successors and assigns of any of the foregoing (the "Spinco Indemnified Parties") from and against any Losses incurred by the Spinco Indemnified Parties (in their capacities as such) to the extent arising out of or resulting from:
 - (i) the conduct of the LIC Retained Businesses (whether before or after the Closing);
 - (ii) the LIC Retained Assets;
 - (iii) the LIC Retained Liabilities (whether incurred before or after the Closing); or
 - (iv) any breach of, or failure to perform or comply with, any covenant, undertaking or obligation of LIC or any of its Subsidiaries (other than the Spinco Entities) under this Agreement, any Restructuring Agreement or any Other Agreement.
- (c) The indemnification provisions set forth in Sections 4.1(a) and (b) shall not apply to: (i) any Losses the responsibility for which is expressly covered by the Tax Sharing Agreement; (ii) any Losses incurred by any Spinco Entity pursuant to any contractual obligation (other than this Agreement, the Restructuring Agreements or the Other Agreements) existing on or after the Closing Date between (x) LIC or any of its Subsidiaries or Affiliates, on the one hand, and (y) Spinco or any of its Subsidiaries or Affiliates, on the other hand; and (iii) any Losses incurred by any LIC Entity pursuant to any contractual obligation (other than this Agreement, the Restructuring Agreements or the Other Agreements) existing on or after the Closing Date between (x) LIC or any of its Subsidiaries or Affiliates, on the one hand, and (y) Spinco or any of its Subsidiaries or Affiliates, on the other hand.
- (d) (i) In connection with any indemnification provided for in this Section 4.1, the party seeking indemnification (the "<u>Indemnitee</u>") will give the party from which indemnification is sought (the "<u>Indemnitor</u>") prompt notice whenever it comes to the attention of the Indemnitee that the Indemnitee has suffered or incurred, or may suffer or incur, any Losses for which it is entitled to indemnification under this Section 4.1, and, if and when known, the facts constituting the basis for such claim and the projected amount of such Losses (which shall not be conclusive as to the amount of such Losses), in each case in reasonable detail. Without limiting the generality of the foregoing, in the case of any Action commenced by a third party for which indemnification is being sought (a "<u>Third-Party Claim</u>"), such notice will be given no later than ten business days following receipt by the Indemnitee of written notice of such Third-Party Claim. Failure by any Indemnitee to so notify the Indemnitor will not affect the rights of such Indemnitee hereunder except to the extent that such failure has a material prejudicial effect on

the defenses or other rights available to the Indemnitor with respect to such Third Party Claim. The Indemnitee will deliver to the Indemnitor as promptly as practicable, and in any event within five business days after Indemnitee's receipt, copies of all notices, court papers and other documents received by the Indemnitee relating to any Third-Party Claim.

- After receipt of a notice pursuant to Section 4.1(d)(i) with respect to any Third-Party Claim, the Indemnitor will be entitled, if it so elects, to take control of the defense and investigation with respect to such Third-Party Claim and to employ and engage attorneys reasonably satisfactory to the Indemnitee to handle and defend such claim, at the Indemnitor's cost, risk and expense, upon written notice to the Indemnitee of such election, which notice acknowledges the Indemnitor's obligation to provide indemnification under this Agreement with respect to any Losses arising out of or relating to such Third-Party Claim. The Indemnitor will not settle any Third-Party Claim that is the subject of indemnification without the written consent of the Indemnitee, which consent will not be unreasonably withheld, conditioned or delayed; provided, however, that, after reasonable notice, the Indemnitor may settle a claim without the Indemnitee's consent if such settlement (A) makes no admission or acknowledgment of Liability or culpability with respect to the Indemnitee, (B) includes a complete release of the Indemnitee and (C) does not seek any relief against the Indemnitee other than the payment of money damages to be borne by the Indemnitor. The Indemnitee will cooperate in all reasonable respects with the Indemnitor and its attorneys in the investigation, trial and defense of any lawsuit or action with respect to such Claim and any appeal arising therefrom (including the filing in the Indemnitee's name of appropriate cross-claims and counterclaims). The Indemnitee may, at its own cost, participate in any investigation, trial and defense of any Third-Party Claim controlled by the Indemnitor and any appeal arising therefrom, including participating in the process with respect to the potential settlement or compromise thereof. If the Indemnitee has been advised by its counsel that there may be one or more legal defenses available to the Indemnitee that conflict with those available to, or that are not available to, the Indemnitor ("Separate <u>Legal Defenses</u>"), or that there may be actual or potential differing or conflicting interests between the Indemnitor and the Indemnitee in the conduct of the defense of such Third-Party Claim, the Indemnitee will have the right, at the expense of the Indemnitor, to engage separate counsel reasonably acceptable to the Indemnitor to handle and defend such Third-Party Claim, provided, that, if such Third-Party Claim can be reasonably separated between those portion(s) for which Separate Legal Defenses are available ("Separable Claims") and those for which no Separate Legal Defenses are available, the Indemnitee will instead have the right, at the expense of the Indemnitor, to engage separate counsel reasonably acceptable to the Indemnitor to handle and defend the Separable Claims, and the Indemnitor will not have the right to control the defense or investigation of such Separable Claims (and, in which case, the Indemnitor will have the right to control the defense or investigation of the remaining portion(s) of such Third-Party Claim).
- (iii) If, after receipt of a notice pursuant to Section 4.1(d)(i) with respect to any Third-Party Claim as to which indemnification is available hereunder, the Indemnitor does not undertake to defend the Indemnitee against such Third-Party Claim, whether by not giving the Indemnitee timely notice of its election to so defend or otherwise, the Indemnitee may, but will have no obligation to, assume its own defense, at the expense of the Indemnitor (including attorneys fees and costs), it being understood that the Indemnitee's right to indemnification for such Third Party Claim shall not be adversely affected by its assuming the defense of such Third Party Claim. The Indemnitor will be bound by the result obtained with respect thereto by the Indemnitee; provided, that the Indemnitee may not settle any lawsuit or action with respect to which the Indemnitee is entitled to indemnification hereunder without the consent of the

Indemnitor, which consent will not be unreasonably withheld, conditioned or delayed; *provided further*, that such consent shall not be required if (i) the Indemnitor had the right under this Section 4.1 to undertake control of the defense of such Third-Party Claim and, after notice, failed to do so within thirty days of receipt of such notice (or such lesser period as may be required by court proceedings in the event of a litigated matter), or (ii) (x) the Indemnitor does not have the right to control the defense of the entirety of such Third-Party Claim pursuant to Section 4.1(d)(ii) or (y) the Indemnitor does not have the right to control the defense of any Separable Claim pursuant to Section 4.1(d)(ii) (in which case such settlement may only apply to such Separable Claims), the Indemnitee provides reasonable notice to Indemnitor of the settlement, and such settlement (A) makes no admission or acknowledgment of Liability or culpability with respect to the Indemnitor, (B) does not seek any relief against the Indemnitor and (C) does not seek any relief against the Indemnitor for money damages.

- (e) In no event will the Indemnitor be liable to any Indemnitee for any special, consequential, indirect, collateral, incidental or punitive damages, however caused and on any theory of liability arising in any way out of this Agreement, whether or not such Indemnitor was advised of the possibility of any such damages; provided, that the foregoing limitations shall not limit a party's indemnification obligations for any Losses incurred by an Indemnitee as a result of the assertion of a Third Party Claim.
- (f) The Indemnitor and the Indemnitee shall use commercially reasonable efforts to avoid production of confidential information, and to cause all communications among employees, counsel and others representing any party with respect to a Third Party Claim to be made so as to preserve any applicable attorney-client or work-product privilege.
- (g) The Indemnitor shall pay all amounts payable pursuant to this Section 4.1 by wire transfer of immediately available funds, promptly following receipt from an Indemnitee of a bill, together with all accompanying reasonably detailed backup documentation, for any Losses that are the subject of indemnification hereunder, unless the Indemnitor in good faith disputes the amount of such Losses or whether such Losses are covered by the Indemnitor's indemnification obligation in which event the Indemnitor shall promptly so notify the Indemnitee. In any event, the Indemnitor shall pay to the Indemnitee, by wire transfer of immediately available funds, the amount of any Losses for which it is liable hereunder no later than three (3) days following any final determination of the amount of such Losses and the Indemnitor's liability therefor. A "final determination" shall exist when (a) the parties to the dispute have reached an agreement in writing or (b) a court of competent jurisdiction shall have entered a final and non-appealable order or judgment.
- (h) If the indemnification provided for in this Section 4.1 shall, for any reason, be unavailable or insufficient to hold harmless an Indemnitee in respect of any Losses for which it is entitled to indemnification hereunder, then the Indemnitor shall contribute to the amount paid or payable by such Indemnitee as a result of such Losses, in such proportion as shall be appropriate to reflect the relative benefits received by and the relative fault of the Indemnitor on the one hand and the Indemnitee on the other hand with respect to the matter giving rise to such Losses.
- (i) The remedies provided in this Section 4.1 shall be cumulative and shall not preclude assertion by any Indemnitee of any other rights or the seeking of any and all other remedies against an Indemnitor, subject to Section 4.1(e).

- (j) The rights and obligations of the LIC Indemnified Persons and the Spinco Indemnified Persons under this Section 4.1 shall survive the Spin-Off.
- (k) For the avoidance of doubt, the provisions of this Section 4.1 are not intended to, and shall not, apply to any Loss, claim or Liability to which the provisions of the Tax Sharing Agreement are applicable.
- (l) To the fullest extent permitted by applicable law, the Indemnitor will indemnify the Indemnitee against any and all reasonable fees, costs and expenses (including attorneys' fees), incurred in connection with the enforcement of his, her or its rights under this Section 4.1.

4.2 Financing

.

- (a) Prior to the Closing, Spinco and one or more of its Subsidiaries shall enter into two or more margin loan agreements (the "Margin Loan Agreements") with certain lenders pursuant to which Spinco and one or more of its Subsidiaries will borrow term loans in a principal amount of \$400 million, secured by the TripAdvisor Securities and guaranteed by Spinco (the "Margin Loans").
- (b) Prior to the Closing, Spinco covenants and agrees, and LIC covenants and agrees to cause its Subsidiary Liberty Interactive LLC, to enter into the Intercompany Note, under which Spinco may request, and Liberty Interactive LLC will agree to fund and advance, from time to time, up to \$200 million in aggregate principal amount of loans to Spinco under the limited circumstances set forth therein.

4.3 Further Assurances

. At any time before or after the Closing, each party hereto covenants and agrees to make, execute, acknowledge and deliver such instruments, agreements, consents, assurances and other documents, and to take all such other commercially reasonable actions, as any other party may reasonably request and as may reasonably be required in order to carry out the purposes and intent of this Agreement and to implement the terms hereof.

4.4 Specific Performance

. Each party hereby acknowledges that the benefits to the other party of the performance by such party of its obligations under this Agreement are unique and that the other party is willing to enter into this Agreement only in reliance that such party will perform such obligations, and agrees that monetary damages may not afford an adequate remedy for any failure by such party to perform any of such obligations. Accordingly, each party hereby agrees that the other party will have the right to enforce the specific performance of such party's obligations hereunder and irrevocably waives any requirement for securing or posting of any bond or other undertaking in connection with the obtaining by the other party of any injunctive or other equitable relief to enforce their rights hereunder.

4.5 Access to Information

.

(a) Each party will provide to the other party, at any time before or after the Distribution Date, upon written request and promptly after the request therefor (subject in all cases, to any bona fide concerns of attorney-client or work-product privilege that any party may reasonably have and any restrictions contained in any agreements or contracts to which any party or its Subsidiaries is a party (it being understood that each of LIC and Spinco will use its reasonable best efforts to provide any such information in a manner that does not result in a violation of a privilege)), any information in its possession or under its control that the

requesting party reasonably needs (i) to comply with reporting, filing or other requirements imposed on the requesting party by a foreign or U.S. federal, state or local judicial, regulatory or administrative authority having jurisdiction over the requesting party or its Subsidiaries, (ii) to enable the requesting party to institute or defend against any action, suit or proceeding in any foreign or U.S. federal, state or local court or (iii) to enable the requesting party to implement the transactions contemplated hereby, including but not limited to performing its obligations under this Agreement, the Restructuring Agreements and the Other Agreements.

- (b) Any information belonging to a party that is provided to another party pursuant to Section 4.5(a) will remain the property of the providing party. The parties agree to cooperate in good faith to take all reasonable efforts to maintain any legal privilege that may attach to any information delivered pursuant to this Section 4.5 or which otherwise comes into the receiving party's possession and control pursuant to this Agreement. Nothing contained in this Agreement will be construed as granting or conferring license or other rights in any such information.
- (c) The party requesting any information under this Section 4.5 will reimburse the providing party for the reasonable out of pocket costs, if any, of creating, gathering and copying such information, to the extent that such costs are incurred for the benefit of the requesting party. No party will have any Liability to any other party if any information exchanged or provided pursuant to this Agreement that is an estimate or forecast, or is based on an estimate or forecast, is found to be inaccurate, absent willful misconduct or fraud by the party providing such information.
- (d) For the avoidance of doubt, the provisions of this Section 4.5 are not intended to, and shall not, apply to any information relating to matters governed by the Tax Sharing Agreement, which shall be subject to the provisions thereof in lieu of this Section 4.5.

4.6 Confidentiality

- . Each party will keep confidential for five years following the Closing Date (or for three years following disclosure to such party, whichever is longer), and will use reasonable efforts to cause its officers, directors, members, employees, Affiliates and agents to keep confidential during such period, all Proprietary Information of the other party, in each case to the extent permitted by applicable law.
- (a) "Proprietary Information" means any proprietary ideas, plans and information, including information of a technological or business nature, of a party (in this context, the "Disclosing Party") (including all trade secrets, intellectual property, data, summaries, reports or mailing lists, in whatever form or medium whatsoever, including oral communications, and however produced or reproduced), that is marked proprietary or confidential, or that bears a marking of like import, or that the Disclosing Party states is to be considered proprietary or confidential, or that a reasonable and prudent person would consider proprietary or confidential under the circumstances of its disclosure. Without limiting the foregoing, all information of the types referred to in the immediately preceding sentence to the extent used by Spinco or the Spinco Businesses or which constitute Spinco Assets on or prior to the Closing Date will constitute Proprietary Information of Spinco for purposes of this Section 4.6.
- (b) Anything contained herein to the contrary notwithstanding, information of a Disclosing Party will not constitute Proprietary Information (and the other party (in this context, the "Receiving Party") will have no obligation of confidentiality with respect thereto), to the

extent such information: (i) is in the public domain other than as a result of disclosure made in breach of this Agreement or breach of any other agreement relating to confidentiality between the Disclosing Party and the Receiving Party; (ii) was lawfully acquired by the Disclosing Party from a third party not bound by a confidentiality obligation; (iii) is approved for release by prior written authorization of the Disclosing Party, or (iv) is disclosed in order to comply with a judicial order issued by a court of competent jurisdiction, or to comply with the laws or regulations of any governmental authority having jurisdiction over the Receiving Party, in which event the Receiving Party will give prior written notice to the Disclosing Party of such disclosure as soon as or to the extent practicable and will cooperate with the Disclosing Party in using reasonable efforts to disclose the least amount of such information required and to obtain an appropriate protective order or equivalent, and provided that the information will continue to be Proprietary Information to the extent it is covered by a protective order or equivalent or is not so disclosed.

4.7 Notices Regarding Transferred Assets

. Any transferor of an Asset or Liability in the Restructuring that receives a notice or other communication from any third party, or that otherwise becomes aware of any fact or circumstance, after the Restructuring, relating to such Asset or Liability, will use commercially reasonable efforts to promptly forward the notice or other communication to the transferee thereof or give notice to such transferee of such fact or circumstance of which it has become aware. The parties will cause their respective Subsidiaries to comply with this Section 4.7.

4.8 Treatment Of Payments

. The parties agree to treat all payments made pursuant to this Agreement in accordance with Section 4.4 of the Tax Sharing Agreement and to increase or reduce any amount paid hereunder if such payment would have been required to be increased or reduced under such section if it were a payment made pursuant to the Tax Sharing Agreement.

ARTICLE V CLOSING

5.1 Closing

. Unless this Agreement is terminated and the transactions contemplated by this Agreement abandoned pursuant to the provisions of Article VI, and subject to the satisfaction or waiver of all conditions set forth in each of Sections 2.2 and 5.2, the closing of the Distribution (the "Closing") will take place at the offices of LIC, at 12300 Liberty Boulevard, Englewood, Colorado, at a mutually acceptable time and date to be determined by LIC (the "Closing Date").

5.2 Conditions to Closing

- .
- (a) The obligations of the parties to complete the transactions provided for herein are conditioned upon the satisfaction or, if applicable, waiver of the conditions set forth in Section 2.2.
 - (b) The performance by each party of its obligations hereunder is further conditioned upon:
 - (i) the performance in all material respects by the other party of its covenants and agreements contained herein to the extent such are required to be performed at or prior to the Closing; and

(ii) the representations and warranties of the other party being true and complete in all material respects as of the Closing Date with the same force and effect as if made at and as of the Closing Date.

5.3 Deliveries at Closing

- (a) <u>LIC</u>. At the Closing, LIC will deliver or cause to be delivered to Spinco:
 - (i) the Tax Sharing Agreement duly executed by an authorized officer of LIC;
- (ii) the Assignment and Assumption Agreement relating to the Governance Agreement duly executed by an authorized officer of LIC;
 - (iii) the BuySeasons Services Agreement duly executed by an authorized officer of Evite, Inc.;
 - (iv) the Services Agreement duly executed by an authorized officer of LMC;
- (v) the Facilities Sharing Agreement duly executed by an authorized officer of Liberty Property Holdings, Inc.;
- (vi) each Aircraft Time Sharing Agreement duly executed by an officer of LMC or one or more of its Subsidiaries, as applicable;
 - (vii) the Intercompany Note duly executed by an authorized officer of Liberty Interactive LLC;
- (viii) a secretary's certificate certifying that the LIC Board has authorized the execution, delivery and performance by LIC of this Agreement, the Restructuring Agreements and the Other Agreements, which authorization will be in full force and effect at and as of the Closing; and
 - (ix) such other documents and instruments as Spinco may reasonably request.
- (b) <u>Spinco</u>. At the Closing, Spinco will deliver or cause to be delivered to LIC:
 - (i) the Tax Sharing Agreement duly executed by an authorized officer of Spinco;
 - (ii) the BuySeasons Services Agreement duly executed by an authorized officer of BuySeasons;
 - (iii) the Services Agreement duly executed by an authorized officer of Spinco;
 - (iv) the Facilities Sharing Agreement duly executed by an authorized officer of Spinco;
- (v) the Assignment and Assumption Agreement relating to the Governance Agreement duly executed by an authorized officer of Spinco;

- (vi) each Aircraft Time Sharing Agreement duly executed by an authorized officer of Spinco;
- (vii) the Margin Loan Agreements duly executed by Spinco, a newly-formed, wholly-owned special purpose subsidiary of Spinco and the financial counterparties thereto:
 - (viii) the Intercompany Note duly executed by an authorized officer of Spinco;
- (ix) a secretary's certificate certifying that the Spinco Board has authorized the execution, delivery and performance by Spinco of this Agreement, the Restructuring Agreements and the Other Agreements, which authorizations will be in full force and effect at and as of the Closing; and
 - (x) such other documents and instruments as LIC may reasonably request.

ARTICLE VI TERMINATION

6.1 Termination

. This Agreement may be terminated and the transactions contemplated hereby may be amended, modified, supplemented or abandoned at any time prior to the Effective Time by and in the sole and absolute discretion of LIC without the approval of Spinco. For the avoidance of doubt, from and after the Effective Time, this Agreement may not be terminated (or any provision hereof modified, amended or waived) without the written agreement of all the parties.

6.2 Effect of Termination

. In the event of any termination of this Agreement in accordance with Section 6.1, this Agreement will immediately become void and the parties will have no Liability whatsoever to each other with respect to the transactions contemplated hereby.

ARTICLE VII MISCELLANEOUS

7.1 Definitions

- (a) For purposes of this Agreement, the following terms have the corresponding meanings:
- "Action" means any demand, action, claim, suit, countersuit, litigation, arbitration, prosecution, proceeding (including any civil, criminal, administrative, investigative or appellate proceeding), hearing, inquiry, audit, examination or investigation whether or not commenced, brought, conducted or heard by or before, or otherwise involving, any court, grand jury or other governmental authority or any arbitrator or arbitration panel.

"Affiliates" means with respect to any Person, any other Person that directly or indirectly, through one or more intermediaries, Controls, is Controlled by, or is under common Control with, such first Person; provided, that, for any purpose hereunder, in each case both before and after the Effective Time, none of the Persons listed in clause (i), (ii), (iii) or (iv) shall be deemed to be Affiliates of any Person listed in any other such

clause: (i) LIC taken together with its Subsidiaries and any of their respective Investees, (ii) Spinco taken together with its Subsidiaries and any of their respective Investees, (iii) LMC taken together with its Subsidiaries and their respective Investees, and (iv) Starz taken together with its Subsidiaries and any of their respective Investees.

"Aircraft Time Sharing Agreement" means the Aircraft Time Sharing Agreements to be entered into between LMC and Spinco, one for each of the two aircraft owned by LMC, substantially in the form attached hereto as Exhibit A-1 and the Aircraft Time Sharing Agreement to be entered into among Subsidiaries of LMC and Spinco for the NetJets aircraft in which such Subsidiaries of LMC have ownership or management rights, as applicable, substantially in the form attached hereto as Exhibit A-2.

"Assets" means assets, properties, interests and rights (including goodwill), wherever located, whether real, personal or mixed, tangible or intangible, movable or immovable, in each case whether or not required by GAAP to be reflected in financial statements or disclosed in the notes thereto.

"<u>BuySeasons Services Agreement</u>" means the services agreement to be entered into between BuySeasons, Inc. and Evite, Inc., substantially in the form attached hereto as <u>Exhibit F</u>.

"Contribution" has the meaning given to such term in the Restructuring Plan.

"Control" means, with respect to any Person, the possession, directly or indirectly, of the power to direct or cause the direction of the management or policies of such Person, whether through ownership of securities or partnership, membership, limited liability company, or other ownership interests, by contract or otherwise and the terms "Controlling" and "Controlled" have meanings correlative to the foregoing.

"DGCL" means the Delaware General Corporation Law.

"Effective Time" means the time at which the Distribution will be effective.

"Exchange Act" means the Securities Exchange Act of 1934, as amended, together with the rules and regulations promulgated thereunder.

"<u>Facilities Sharing Agreement</u>" means the Facilities Sharing Agreement to be entered into between Liberty Property Holdings, Inc. and Spinco, substantially in the form attached hereto as <u>Exhibit B.</u>

"GAAP" means generally accepted accounting principles as in effect from time to time in the United States, consistently applied.

"Governance Agreement" means the Governance Agreement, dated as of December 20, 2011, by and between TripAdvisor and LIC.

"Governmental Authorization" means any authorization, approval, consent, license, certificate or permit issued, granted, or otherwise made available under the authority of any court, governmental or regulatory authority, agency, stock exchange, commission or body.

- "Intercompany Note" means the master promissory note to be entered into between Spinco and Liberty Interactive LLC, a Subsidiary of LIC.
 - "IRS" means the Internal Revenue Service.
- "Investee" of any Person means any Person in which such first Person owns or controls an equity or voting interest.
- "<u>Liabilities</u>" means any and all debts, liabilities, commitments and obligations, whether or not fixed, contingent or absolute, matured or unmatured, direct or indirect, liquidated or unliquidated, accrued or unaccrued, known or unknown, and whether or not required by GAAP to be reflected in financial statements or disclosed in the notes thereto (other than taxes).
 - "LIC Board" means the Board of Directors of LIC or a duly authorized committee thereof.
- "<u>LIC Charter"</u> means the Restated Certificate of Incorporation of LIC, as in effect immediately prior to the Distribution Date.
 - "LIC Common Stock" means LINTA, LINTB, LVNTA and LVNTB.
- "LIC Entity" or "LIC Entities" means and includes each of LIC and its Subsidiaries (other than the Spinco Entities), after giving effect to the Restructuring.
 - "LIC Retained Assets" means all Assets which are held at the Effective Time by L IC.
 - "LIC Retained Businesses" means all businesses which are held at the Effective Time by L IC.
 - "LIC Retained Liabilities" means all Liabilities which are held at the Effective time by LIC.
 - "Liberty Interactive Common Stock" means LINTA and LINTB.
- "<u>Liberty Ventures Option</u>" means an option to purchase shares of Liberty Ventures Common Stock pursuant to a stock incentive plan of LIC.
- "<u>Liberty Ventures SAR</u>" means a stock appreciation right with respect to shares of Liberty Ventures Common Stock granted under a stock incentive plan of LIC.
 - "LINTA" means LIC's Series A Liberty Interactive common stock, par value \$.01 per share.
 - "LINTB" means LIC's Series B Liberty Interactive common stock, par value \$.01 per share.
 - "LMC" means Liberty Media Corporation.

"Losses" means any and all damages, losses, deficiencies, Liabilities, penalties, judgments, settlements, claims, payments, fines, interest, costs and expenses (including the fees and expenses of any and all actions and demands, assessments, judgments, settlements and compromises relating thereto and the costs and expenses of attorneys', accountants', consultants' and other professionals' fees and expenses incurred in the investigation or defense thereof or in asserting, preserving or enforcing an Indemnitee's rights hereunder), whether in connection with a Third-Party Claim or otherwise.

"Order" means any order, injunction, judgment, decree or ruling of any court, governmental or regulatory authority, agency, commission or body.

"Person" means any individual, corporation, company, partnership, trust, incorporated or unincorporated association, joint venture or other entity of any kind.

"Qualifying Subsidiary" means a former direct or indirect Subsidiary of LIC, any successor of any such former Subsidiary, and the parent company (directly or indirectly) of any such former Subsidiary or successor, including Spinco, LMC, Ascent Capital Group, Inc., Discovery Communications, Inc., Liberty Global, Inc. and Starz.

"Restructuring Plan" means the Restructuring Plan attached hereto as Schedule 1.1.

"SEC" means the Securities and Exchange Commission.

"SEC Filings" means the Registration Statement or any amendments or supplements thereto, including any preliminary filings of the same, and any other registration statements or reports filed under the Securities Act or Exchange Act, in connection with the Spin-Off.

"Securities Act" means the Securities Act of 1933, as amended, together with all rules and regulations promulgated thereunder.

"Services Agreement" means the Services Agreement to be entered into between LMC and Spinco, substantially in the form attached hereto as Exhibit C.

"Spinco Assets" means BuySeasons, Inc. and the TripAdvisor Securities and all Assets related thereto.

"Spinco Board" means the Board of Directors of Spinco or a duly authorized committee thereof.

"Spinco Businesses" means Spinco's consolidated Subsidiary TripAdvisor and Spinco's wholly owned Subsidiary BuySeasons, Inc.

"Spinco Charter" means the Restated Certificate of Incorporation of Spinco to be filed with the Delaware Secretary of State immediately prior to the Effective Time, substantially in the form attached hereto as Exhibit D.

"Spinco Entity" or "Spinco Entities" means and includes each of Spinco and its Subsidiaries, after giving effect to the Restructuring.

"Spinco Liabilities" means all Liabilities of LIC relating to BuySeasons, Inc. and to the TripAdvisor Securities.

"Spinco Option" means any option to purchase shares of Spinco Common Stock issued pursuant to the Spinco Transitional Plan.

"Spinco Restricted Stock" means any shares of Spinco Common Stock subject to restricted stock awards issued pursuant to the Spinco Transitional Plan.

"Spinco SARs" means stock appreciation rights with respect to Spinco Common Stock issued pursuant to the Spinco Transitional Plan.

"Spinco Common Stock" means the Series A and Series B common stock, par value \$.01 per share, of Spinco.

"Spinco Transitional Plan" means the Liberty TripAdvisor Holdings, Inc. Transitional Stock Adjustment Plan.

"Subsidiary" when used with respect to any Person, means (i)(A) a corporation a majority in voting power of whose share capital or capital stock with voting power, under ordinary circumstances, to elect directors is at the time, directly or indirectly, owned by such Person, by one or more Subsidiaries of such Person, or by such Person and one or more Subsidiaries of such Person, whether or not such power is subject to a voting agreement or similar encumbrance, (B) a partnership or limited liability company in which such Person or a Subsidiary of such Person is, at the date of determination, (1) in the case of a partnership, a general partner of such partnership with the power affirmatively to direct the policies and management of such partnership or (2) in the case of a limited liability company, the managing member or, in the absence of a managing member, a member with the power affirmatively to direct the policies and management of such limited liability company, or (C) any other Person (other than a corporation) in which such Person, one or more Subsidiaries of such Person or such Person and one or more Subsidiaries of such Person, directly or indirectly, at the date of determination thereof, has or have (1) the power to elect or direct the election of a majority of the members of the governing body of such Person, whether or not such power is subject to a voting agreement or similar encumbrance, or (2) in the absence of such a governing body, at least a majority ownership interest or (ii) any other Person of which an aggregate of 50% or more of the equity interests are, at the time, directly or indirectly, owned by such Person and/or one or more Subsidiaries of such Person. For purposes of this Agreement, both prior to and after the Effective Time, none of Spinco and its Subsidiaries shall be deemed to be Subsidiaries of LIC or any of its Subsidiaries.

"<u>Tax Sharing Agreement</u>" means the Tax Sharing Agreement to be entered into between LIC and Spinco, substantially in the form attached hereto as <u>Exhibit E</u>.

"Treasury Regulations" means the Treasury regulations promulgated under the Code.

" $\underline{\text{TripAdvisor Securities}}$ " means the shares of common stock, par value \$0.001 per share, and Class B common stock, par value \$0.001 per share, in each case, of

TripAdvisor, beneficially owned by LIC immediately prior to the Effective Time of the Spin-Off.

(b) As used herein, the following terms will have the meanings set forth in the applicable section of this Agreement set forth below:

Adjusted Liberty Ventures Option Section 2.3(b)(i) Adjusted Liberty Ventures SAR Section 2.3(c)(i) Agreement Preamble Awards Section 2.3(a) Closing Section 5.1 Closing Date Section 5.1 Code Recitals **Disclosing Party** Section 4.6(a) Distribution Recitals Distribution Date Section 2.1(a) Indemnitee Section 4.1(d)(i) Section 4.1(d)(i) Indemnitor Liberty Ventures Common Stock Recitals Liberty Ventures Record Holders Section 2.1(b) Liberty Ventures Restricted Stock Section 2.3(d) LIC Preamble LIC Indemnified Parties Section 4.1(a) LVNTA Recitals LVNTB Recitals Margin Loan Agreements Section 4.2(a) Margin Loans Section 4.2(a) Outstanding Liberty Ventures Option Section 2.3(b)(i) Outstanding Liberty Ventures SAR Section 2.3(c)(i) Other Agreements Section 3.1(b) Post Spin Awards Section 2.3(g) **Proprietary Information** Section 4.6(a) Receiving Party Section 4.6(b) Record Date Section 2.1(a) Registration Statement Section 2.2(d) Restructuring Section 1.1(a) Restructuring Agreements Section 3.1(b) Ruling Section 2.2(b) Separable Claims Section 4.1(d)(ii) Separate Legal Defenses Section 4.1(d)(ii) Spin-Off Recitals Spinco Preamble Spinco Option Section 2.3(b)(i) Spinco SAR Section 2.3(c)(i) Spinco Common Stock Section 2.1(b) Spinco Indemnified Parties Section 4.1(b) Spinco Restricted Stock Section 2.3(d) Spinco Series A Common Stock Section 2.1(b)

Spinco Series B Common Stock Third-Party Claim TripAdvisor Section 2.1(b) Section 4.1(d)(i) Recitals

7.2 No Third-Party Rights

. Except for the indemnification rights of the LIC Indemnified Persons and the Spinco Indemnified Persons pursuant to Section 4.1, nothing expressed or referred to in this Agreement is intended or will be construed to give any Person other than the parties hereto and their respective successors and assigns any legal or equitable right, remedy or claim under or with respect to this Agreement, or any provision hereof, it being the intention of the parties hereto that this Agreement and all of its provisions and conditions are for the sole and exclusive benefit of the parties to this Agreement and their respective successors and assigns.

7.3 Notices

. All notices and other communications hereunder shall be in writing and shall be delivered in person, by facsimile (with confirming copy sent by one of the other delivery methods specified herein), by overnight courier or sent by certified, registered or express air mail, postage prepaid, and shall be deemed given when so delivered in person, or when so received by facsimile or courier, or, if mailed, three (3) calendar days after the date of mailing, as follows:

if to any LIC Entity: Liberty Interactive Corporation

12300 Liberty Boulevard Englewood, Colorado 80112 Facsimile (720) 875-5401 Attention: General Counsel

if to any Spinco Entity: Liberty TripAdvisor Holdings, Inc.

12300 Liberty Boulevard Englewood, Colorado 80112 Facsimile (720) 875-5401 Attention: General Counsel

or to such other address as the party to whom notice is given may have previously furnished to the other party in writing in the manner set forth above.

7.4 Entire Agreement

. This Agreement (including the Exhibits and Schedules attached hereto) together with the Restructuring Agreements and the Other Agreements (including the Tax Sharing Agreement) embodies the entire understanding among the parties relating to the subject matter hereof and thereof and supersedes and terminates any prior agreements and understandings among the parties with respect to such subject matter, and no party to this Agreement shall have any right, responsibility or Liability under any such prior agreement or understanding. Any and all prior correspondence, conversations and memoranda are merged herein and shall be without effect hereon. No promises, covenants or representations of any kind, other than those expressly stated herein, have been made to induce either party to enter into this Agreement.

7.5 Binding Effect; Assignment

. This Agreement and all of the provisions hereof shall be binding upon and inure to the benefit of the parties hereto and their respective successors and

permitted assigns. Except with respect to a merger of a party, neither this Agreement nor any of the rights, interests or obligations hereunder shall be assigned by any party hereto without the prior written consent of the other parties; *provided*, *however*, that LIC and Spinco may assign their respective rights, interests, duties, liabilities and obligations under this Agreement to any of their respective wholly-owned Subsidiaries, but such assignment shall not relieve LIC or Spinco, as the assignor, of its obligations hereunder.

7.6 Governing Law; Jurisdiction

. This Agreement and the legal relations among the parties hereto will be governed in all respects, including validity, interpretation and effect, by the laws of the State of Delaware applicable to contracts made and performed wholly therein, without giving effect to any choice or conflict of laws provisions or rules that would cause the application of the laws of any other jurisdiction. Each of the parties hereto irrevocably agrees that any legal action or proceeding with respect to this Agreement, and the rights and obligations arising hereunder, or for recognition and enforcement of any judgment in respect of this Agreement, and the rights and obligations arising hereunder brought by the other party hereto or its successors or assigns, shall be brought and determined exclusively in the Delaware Court of Chancery and any state appellate court therefrom within the State of Delaware (or, if the Delaware Court of Chancery declines to accept jurisdiction over a particular matter, any state or federal court within the State of Delaware). Each of the parties hereto hereby irrevocably submits with regard to any such action or proceeding for itself and in respect of its property, generally and unconditionally, to the personal jurisdiction of the aforesaid courts and agrees that it will not bring any action relating to this Agreement or the transactions contemplated hereby in any court other than the aforesaid courts. Each of the parties hereto hereby irrevocably waives, and agrees not to assert as a defense, counterclaim or otherwise, in any action or proceeding with respect to this Agreement (a) any claim that it is not personally subject to the jurisdiction of the above named courts for any reason other than the failure to serve in accordance with Section 7.3 and this Section 7.6, (b) any claim that it or its property is exempt or immune from jurisdiction of any such court or from any legal process commenced in such courts (whether through service of notice, attachment prior to judgment, attachment in aid of execution of judgment, execution of judgment or otherwise) and (c) to the fullest extent permitted by applicable law, any claim that (i) the suit, action or proceeding in such court is brought in an inconvenient forum, (ii) the venue of such suit, action or proceeding is improper or (iii) this Agreement or the subject matter hereof may not be enforced in or by such courts. Process in any such suit, action or proceeding may be served on any party anywhere in the world, whether within or without the jurisdiction of any such court. Without limiting the foregoing, each party agrees that service of process on such party as provided in Section 7.3 shall be deemed effective service of process on such party.

7.7 Waiver of Jury Trial

. EACH PARTY ACKNOWLEDGES AND AGREES THAT ANY CONTROVERSY WHICH MAY ARISE UNDER THIS AGREEMENT IS LIKELY TO INVOLVE COMPLICATED AND DIFFICULT ISSUES, AND, THEREFORE, EACH PARTY HEREBY IRREVOCABLY AND UNCONDITIONALLY WAIVES TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, ANY RIGHT SUCH PARTY MAY HAVE TO A TRIAL BY JURY IN RESPECT TO ANY ACTION DIRECTLY OR INDIRECTLY ARISING OUT OF, UNDER OR IN CONNECTION WITH OR RELATING TO THIS AGREEMENT. EACH PARTY CERTIFIES AND ACKNOWLEDGES THAT (A) NO REPRESENTATIVE, AGENT OR ATTORNEY OF ANY OTHER PARTY HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH OTHER PARTY WOULD NOT, IN THE EVENT OF SUCH ACTION, SEEK TO ENFORCE THE FOREGOING WAIVER, (B) EACH PARTY UNDERSTANDS AND HAS CONSIDERED THE

IMPLICATIONS OF THIS WAIVER, (C) EACH PARTY MAKES THIS WAIVER VOLUNTARILY, AND (D) EACH PARTY HAS BEEN INDUCED TO ENTER INTO THIS AGREEMENT BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS AND CERTIFICATIONS IN THIS SECTION 7.7.

7.8 Severability

. Any provision of this Agreement which is prohibited or unenforceable in any jurisdiction shall, as to such jurisdiction, be ineffective to the extent of such prohibition or unenforceability without invalidating the remaining provisions hereof. Any such prohibition or unenforceability in any jurisdiction shall not invalidate or render unenforceable such provision in any other jurisdiction. Upon a determination that any provision of this Agreement is prohibited or unenforceable in any jurisdiction, the parties shall negotiate in good faith to modify this Agreement so as to effect the original intent of the parties as closely as possible in an acceptable manner in order that the provisions contemplated hereby are consummated as originally contemplated to the fullest extent possible.

7.9 Amendments; Waivers

. Any provision of this Agreement may be amended or waived if, but only if, such amendment or waiver is in writing and is signed, in the case of an amendment, by each party to this Agreement, or in the case of a waiver, by the party against whom the waiver is to be effective. No failure or delay by any party in exercising any right, power or privilege hereunder shall operate as a waiver thereof nor shall any single or partial exercise thereof preclude any other or further exercise thereof or the exercise of any other right, power or privilege. Except as otherwise provided herein, the rights and remedies herein provided shall be cumulative and not exclusive of any rights or remedies provided by applicable law. Any consent provided under this Agreement must be in writing, signed by the party against whom enforcement of such consent is sought.

7.10 No Strict Construction; Interpretation

.

- (a) LIC and Spinco each acknowledge that this Agreement has been prepared jointly by the parties hereto and shall not be strictly construed against any party hereto.
- When a reference is made in this Agreement to an Article, Section, Exhibit or Schedule, such reference shall be to an Article of, a Section of, or an Exhibit or Schedule to, this Agreement unless otherwise indicated. The table of contents and headings contained in this Agreement are for reference purposes only and shall not affect in any way the meaning or interpretation of this Agreement. Whenever the words "include", "includes" or "including" are used in this Agreement, they shall be deemed to be followed by the words "without limitation." The words "hereof", "herein" and "hereunder" and words of similar import when used in this Agreement shall refer to this Agreement as a whole and not to any particular provision of this Agreement. All terms defined in this Agreement shall have the defined meanings when used in any certificate or other document made or delivered pursuant hereto unless otherwise defined therein. The definitions contained in this Agreement are applicable to the singular as well as the plural forms of such terms and to the masculine as well as to the feminine and neuter genders of such term. Any agreement, instrument or statute defined or referred to herein or in any agreement or instrument that is referred to herein means such agreement, instrument or statute as from time to time amended, modified or supplemented, including (in the case of agreements or instruments) by waiver or consent and (in the case of statutes) by succession of comparable successor statutes and references to all attachments

thereto and instruments incorporated therein. References to a Person are also to its permitted successors and assigns and references to a party means a party to this Agreement.

7.11 Conflicts with Tax Sharing Agreement

. In the event of a conflict between this Agreement and the Tax Sharing Agreement, the provisions of the Tax Sharing Agreement shall prevail.

7.12 Counterparts

. This Agreement may be executed in two or more identical counterparts, each of which shall be deemed to be an original, and all of which together shall constitute one and the same agreement. The Agreement may be delivered by facsimile transmission of a signed copy thereof.

IN WITNESS WHEREOF, the parties hereto have executed this Agreement as of the day and year first above written.

LIBERTY INTERACTIVE CORPORATION

By: <u>/s/ Richard N. Baer</u> Name: Richard N, Baer

Title: Senior Vice President and General Counsel

LIBERTY TRIPADVISOR HOLDINGS, INC.

By: <u>/s/ Pamela L. Coe</u> Name: Pamela L. Coe

Title: Vice President and Secretary

List of Omitted Exhibits and Schedules

The following exhibits and schedules to the Reorganization Agreement, dated as of August 15, 2014, between Liberty Interactive Corporation and Liberty TripAdvisor Holdings, Inc. have not been provided herein:

Exhibit A-1 — Form of Aircraft Time Sharing Agreement (LMC Owned Aircraft) (See Exhibit 10.4 to Liberty TripAdvisor Holdings Inc.'s Current Report on Form 8-K, filed on September 3, 2014 (the "8-K"))

Exhibit A-2 — Form of Aircraft Time Sharing Agreement (NetJets Aircraft) (See Exhibit 10.5 to the 8-K)

Exhibit B — Form of Facilities Sharing Agreement (See Exhibit 10.3 to the 8-K)

Exhibit C — Form of Services Agreement (See Exhibit 10.2 to the 8-K)

Exhibit D — Form of Spinco Charter (See Exhibit 3.1 to the 8-K)

Exhibit E — Form of Tax Sharing Agreement (See Exhibit 10.1 to the Registrant's Current Report on Form 8-K, filed September 3, 2014)

Exhibit F — Form of Services Agreement between BuySeasons, Inc. and Evite, Inc.

Schedule 1.1 — Restructuring Plan

The undersigned registrant hereby undertakes to furnish supplementally a copy of any omitted exhibit or schedule to the Securities and Exchange Commission upon request.

TAX SHARING AGREEMENT

BETWEEN

LIBERTY INTERACTIVE CORPORATION

AND

LIBERTY TRIPADVISOR HOLDINGS, INC.

TABLE OF CONTENTS

		Page
Section 1.	Definition of Terms	3
Section 2.	Allocation of Taxes and Tax-Related Losses	10
2.1	Allocation of Taxes	10
2.2	Special Rules	12
2.3	Tax Payments	12
Section 3.	Preparation and Filing of Tax Returns	12
3.1	Combined Returns	12
3.2	Separate Returns	12
3.3	Provision of Information	13
3.4	Special Rules Relating to the Preparation of Tax Returns	13
3.5	Refunds, Credits or Offsets	15
3.6	Carrybacks	15
3.7	Amended Returns	16
Section 4.	Tax Payments	16
4.1	Payment of Taxes to Tax Authority	16
4.2	Indemnification Payments	16
4.3	Interest on Late Payments	17
4.4	Tax Consequences of Payments	17
Section 5.	Assistance and Cooperation	17
5.1	Cooperation	17
5.2	Supplemental Rulings	18
Section 6.	Tax Records	19
6.1	Retention of Tax Records	19
6.2	Access to Tax Records	19
6.3	Confidentiality	19
6.4	Delivery of Tax Records	20
Section 7.	Restriction on Certain Actions of Distributing and Spinco; Indemnity	20
7.1	Restrictive Covenants	20
7.2	Distributing Indemnity	21
7.3	Spinco Indemnity	21
7.4	Scope	21
7.5	Notices of Tax Contests (Other than Joint Claims)	22
7.6	Control of Tax Contests (Other than Joint Claims)	22
7.7	Cooperation	22
7.8	Joint Claims	22
Section 8.	General Provisions	23
8.1	Termination	23
8.2	Predecessors or Successors	23
8.3	Expenses	23
8.4	Governing Law	23
8.5	Waiver of Jury Trial	24
8.6	Notices	24
8.7	Counterparts	25
8.8	Binding Effect; Assignment	25
8.9	Severability	25

8.10	Amendments; Waivers	25
8.11	Effective Date	26
8.12	Change in Law	26
8.13	Authorization, Etc.	26
8.14	No Third Party Beneficiaries	26
8.15	Entire Agreement	26
8.16	No Strict Construction; Interpretation	26
8.17	Headings	27

TAX SHARING AGREEMENT

THIS TAX SHARING AGREEMENT (this "Agreement") is entered into as of August 27, 2014, between Liberty Interactive Corporation, a Delaware corporation ("Distributing"), and Liberty TripAdvisor Holdings, Inc., a Delaware corporation ("Spinco"). Unless otherwise indicated, all "Section" references in this Agreement are to sections of this Agreement.

RECITALS

WHEREAS, Spinco is a wholly owned subsidiary of Distributing; and

WHEREAS, the Board of Directors of Distributing has determined that it would be appropriate and desirable for Distributing to separate the Spinco Group from the Distributing Group; and

WHEREAS, the Board of Directors of Spinco has also approved such transaction; and

WHEREAS, following the Contribution, Distributing intends to distribute its entire interest in the stock of Spinco to holders of Liberty Ventures Common Stock (the "Distribution"), in what is intended to qualify as a tax-free transaction described under Sections 368(a), 355, and 361 of the Code; and

WHEREAS, the parties set forth in the Reorganization Agreement the principal arrangements between them regarding the separation of the Spinco Group from the Distributing Group; and

WHEREAS, the parties desire to provide for and agree upon the allocation between the parties of liabilities for Taxes arising prior to, as a result of, and subsequent to the Distribution, and to provide for and agree upon other matters relating to Taxes.

NOW, THEREFORE, in consideration of the foregoing and the covenants and agreements set forth below, and intending to be legally bound hereby, Distributing and Spinco hereby agree as follows:

SECTION 1.Definition of Terms

. For purposes of this Agreement (including the recitals hereof), the following terms have the following meanings:

"Affiliate" means with respect to any Person, any other Person that directly or indirectly, through one or more intermediaries, Controls, is Controlled by, or is under common Control with, such first Person. For the avoidance of doubt, (x) neither Expedia, Inc. nor any of its respective Subsidiaries will be treated as Affiliates of any member of the Distributing Group or the Spinco Group; (y) no member of the Spinco Group will be treated as an Affiliate of any member of the Distributing Group; and (z) no member of the Distributing Group will be treated as an Affiliate of any member of the Spinco Group.

"Agreement" has the meaning set forth in the preamble hereof.

"business day" means any day other than a Saturday, Sunday or a day on which banking institutions in New York City, New York or London, England are authorized or required by law or executive order to close.

"Closing of the Books Method" means the apportionment of items between portions of a Taxable period based on a closing of the books and records as of the end of the day on the Distribution Date (as if the Distribution Date were the end of the Taxable period), provided that any items not susceptible to such apportionment shall be apportioned pro rata on the basis of elapsed days during the relevant portion of the Taxable period.

"Code" means the U.S. Internal Revenue Code of 1986, as amended from time to time, or any successor law.

"Combined Return" means a consolidated, combined or unitary Tax Return that includes, by election or otherwise, one or more members of the Distributing Group and one or more members of the Spinco Group.

"Company" means Distributing or Spinco, as the context requires.

"Compensatory Equity Interests" means options, stock appreciation rights, restricted stock, stock units or other rights with respect to Distributing Stock or Spinco Stock that are granted on or prior to the Distribution Date by Distributing, Spinco or any of their respective Subsidiaries in connection with employee, independent contractor or director compensation or other employee benefits (including, for the avoidance of doubt, options, stock appreciation rights, restricted stock, stock units or other rights issued in substitution for any of the foregoing by reason of the Distribution or any subsequent transaction).

"Contribution" has the meaning given to such term in the Reorganization Agreement.

"Control" means, with respect to any Person, the possession, directly or indirectly, of the power to direct or cause the direction of the management or policies of such Person, whether through ownership of securities or partnership, membership, limited liability company, or other ownership interests, by contract or otherwise and the terms "Controlling" and "Controlled" have meanings correlative to the foregoing.

"Delaware Chancery Court" has the meaning set forth in Section 8.4.

"Disclosing Party" has the meaning set forth in Section 6.3.

"Distributing" has the meaning set forth in the preamble hereof.

"Distributing Group" means Distributing and each Subsidiary of Distributing (but only while such Subsidiary is a Subsidiary of Distributing) other than any

Person that is a member of the Spinco Group (but only during the period such Person is treated as a member of the Spinco Group).

"Distributing Indemnitees" has the meaning set forth in Section 7.3.

"Distributing Stock" means (w) the Series A Liberty Interactive common stock, par value \$.01 per share, the Series B Liberty Interactive common stock, par value \$.01 per share, the Series A Liberty Ventures common stock, par value \$.01 per share, and the Series B Liberty Ventures common stock, par value \$.01 per share, (x) if and when the proposed reclassification of Distributing's stock is effected, the Series A QVC Group common stock, par value \$.01 per share, the Series B QVC Group common stock, par value \$.01 per share, the Series A Liberty Digital Commerce common stock, par value \$.01 per share, (y) if and when issued, the Series C Liberty Interactive common stock, par value \$.01 per share, the Series C Liberty Ventures common stock, par value \$.01 per share, the Series C QVC Group common stock, par value \$.01 per share, and the Series C Liberty Digital Commerce common stock, par value \$.01 per share, and (z) any series or class of stock into which the Series A, Series B, or Series C Liberty Interactive common stock, the Series A, Series B, or Series C QVC Group common stock, or the Series A, Series B, or Series C Liberty Digital Commerce common stock is redesignated, reclassified, converted or exchanged following the Effective Time.

"Distribution" has the meaning set forth in the recitals hereof.

"Distribution Date" means the date on which the Distribution occurs.

"Due Date" has the meaning set forth in Section 4.3.

"Effective Time" means the time at which the Distribution is effected on the Distribution Date.

"Employing Party" has the meaning set forth in Section 3.4(d)(i).

"Final Determination" means a determination within the meaning of Section 1313 of the Code or any similar provision of state or local Tax Law.

"Group" means the Distributing Group or the Spinco Group, as the context requires.

"Income Tax" means all Taxes (i) based upon, measured by, or calculated with respect to, net income, net profits or deemed net profits (including any capital gains Tax, minimum Tax based upon, measured by, or calculated with respect to, net income, net profits or deemed net profits, any Tax on items of Tax preference and depreciation recapture or clawback, but not including sales, use, real or personal property, gross or net receipts, gross profits, transfer and similar Taxes), (ii) imposed by a foreign country which qualify under Section 903 of the Code or (iii) based upon, measured by, or calculated with respect to multiple bases (including, but not limited to, corporate franchise and occupation

Taxes) if such Taxes may be based upon, measured by, or calculated with respect to one or more bases described in clause (i) above.

"Interest Rate" means the Rate determined below, as adjusted as of each Interest Rate Determination Date. The "Rate," means, with respect to each period between two consecutive Interest Rate Determination Dates, a rate determined at approximately 11:00 a.m., London time, two London business days before the first Interest Rate Determination Date equal to the greater of: (x) the sum of (i) the six month dollar LIBOR rate as displayed on page "LR" of Bloomberg (or such other appropriate page as may replace such page), plus (ii) 2%, and (y) the interest rate that would be applicable at such time to a "large corporate underpayment" (within the meaning of Section 6621(c) of the Code) under Sections 6601 and 6621 of the Code. Interest will be calculated on the basis of a year of 365 days and the actual number of days for which due.

"Interest Rate Determination Date" means the Due Date and each March 31, June 30, September 30 and December 31 thereafter.

"IRS" means the Internal Revenue Service.

"IRS Submissions" means the Ruling Request, each supplemental submission and any other correspondence or supplemental materials submitted to the IRS in connection with obtaining the Ruling.

"issuing corporation" has the meaning set forth in Section 3.4(d)(ii).

"Joint Claim" has the meaning set forth in Section 7.8.

"Liberty Ventures Common Stock" means the Series A Liberty Ventures common stock, par value \$.01 per share, and the Series B Liberty Ventures common stock, par value \$.01 per share.

"Losses" means any and all damages, losses, deficiencies, liabilities, obligations, penalties, judgments, settlements, claims, payments, fines, interest, costs and expenses (including the fees and expenses of any and all actions and demands, assessments, judgments, settlements and compromises relating thereto and the costs and expenses of attorneys', accountants', consultants' and other professionals' fees and expenses incurred in the investigation or defense thereof or the enforcement of rights hereunder); provided, however, that "Losses" shall exclude any special or punitive damages; provided, further, that the foregoing proviso will not be interpreted to limit indemnification for Losses incurred as a result of the assertion by a claimant (other than the parties hereto and their successors and assigns) in a third-party claim for special or punitive damages.

"Non-Preparer" means the Company that is not responsible for the preparation and filing of the applicable Tax Return pursuant to Sections 3.1 or 3.2.

"Payment Date" means (x) with respect to any U.S. federal income tax return, the due date for any required installment of estimated taxes determined under Code Section 6655, the due date (determined without regard to extensions) for filing the return

determined under Code Section 6072, and the date the return is filed, and (y) with respect to any other Tax Return, the corresponding dates determined under the applicable Tax Law.

"Person" means any individual, corporation, company, partnership, trust, incorporated or unincorporated association, joint venture or other entity of any kind.

"Post-Distribution Period" means any Tax Year or other Taxable period beginning after the Distribution Date and, in the case of any Straddle Period, that part of the Tax Year or other taxable period that begins at the beginning of the day after the Distribution Date.

"Pre-Distribution Period" means any Tax Year or other taxable period that ends on or before the Distribution Date and, in the case of any Straddle Period, that part of the Tax Year or other taxable period through the end of the day on the Distribution Date.

"Preparer" means the Company that is responsible for the preparation and filing of the applicable Tax Return pursuant to Sections 3.1 or 3.2.

"Receiving Party" has the meaning set forth in Section 6.3.

"Reorganization Agreement" means the Reorganization Agreement between Distributing and Spinco dated August 15, 2014.

"Requesting Party" has the meaning set forth in Section 5.2(a).

"Restructuring" has the meaning assigned to such term in the Reorganization Agreement.

"Ruling" means PLR 135954-13 that was issued to Distributing on February 28, 2014.

"Ruling Request" means the request for rulings, dated August 15, 2013, filed by Distributing with the IRS in connection with the Distribution, as the same shall have been amended or supplemented.

"Separate Return" means (a) in the case of any Tax Return required to be filed by any member of the Distributing Group (including any consolidated, combined or unitary Tax Return), any such Tax Return that does not include any member of the Spinco Group, and (b) in the case of any Tax Return required to be filed by any member of the Spinco Group (including any consolidated, combined or unitary Tax Return), any such Tax Return that does not include any member of the Distributing Group.

"Spinco" has the meaning set forth in the preamble hereof.

"Spinco Group" means (x) with respect to any Tax Year (or portion thereof) ending at or before the Effective Time, Spinco and each of its Subsidiaries at the Effective Time; and (y) with respect to any Tax Year (or portion thereof) beginning after the

Effective Time, Spinco and each Subsidiary of Spinco (but only while such Subsidiary is a Subsidiary of Spinco).

"Spinco Indemnitees" has the meaning set forth in Section 7.2.

"Spinco Section 355(e) Event" means the application of Section 355(e) of the Code to the Distribution as a result of the Distribution being "part of a plan (or series of related transactions) pursuant to which 1 or more persons acquire directly or indirectly stock representing a 50-percent or greater interest" in Spinco (within the meaning of Section 355(e) of the Code).

"Spinco Stock" means the Series A Spinco common stock, par value \$.01 per share, the Series B Spinco common stock, par value \$.01 per share, and if and when issued, the Series C Spinco common stock, par value \$.01 per share, and any series or class of stock into which the Series A, Series B, or Series C Spinco common stock is redesignated, reclassified, converted or exchanged following the Effective Time.

"Straddle Period" means any Taxable period commencing on or prior to, and ending after, the Distribution Date.

"Subsidiary" when used with respect to any Person, means (i)(A) a corporation a majority in voting power of whose share capital or capital stock with voting power, under ordinary circumstances, to elect directors is at the time, directly or indirectly, owned by such Person, by one or more Subsidiaries of such Person, or by such Person and one or more Subsidiaries of such Person, whether or not such power is subject to a voting agreement or similar encumbrance, (B) a partnership or limited liability company in which such Person or a Subsidiary of such Person is, at the date of determination, (1) in the case of a partnership, a general partner of such partnership with the power affirmatively to direct the policies and management of such partnership or (2) in the case of a limited liability company, the managing member or, in the absence of a managing member, a member with the power affirmatively to direct the policies and management of such limited liability company, or (C) any other Person (other than a corporation) in which such Person, one or more Subsidiaries of such Person or such Person and one or more Subsidiaries of such Person, directly or indirectly, at the date of determination thereof, has or have (1) the power to elect or direct the election of a majority of the members of the governing body of such Person, whether or not such power is subject to a voting agreement or similar encumbrance, or (2) in the absence of such a governing body, at least a majority ownership interest or (ii) any other Person of which an aggregate of 50% or more of the equity interests are, at the time, directly or indirectly, owned by such Person and/or one or more Subsidiaries of such Person. For purposes of the foregoing, (x) neither Expedia, Inc. nor any of its respective Subsidiaries will be treated as Subsidiaries of Distributing, and (y) neither TripAdvisor nor any of its Subsidiaries will be treated as Subsidiaries of Distributing or Spinco during any period in which such party beneficially owns less than 50% of the outstanding common stock of TripAdvisor.

"Supplemental IRS Submissions" means any request for a Supplemental Ruling, each supplemental submission and any other correspondence or supplemental materials submitted to the IRS in connection with obtaining any Supplemental Ruling.

"Supplemental Ruling" means any private letter ruling obtained by Distributing or Spinco from the IRS which supplements or otherwise modifies the Ruling.

"Tax" or "Taxes" means any net income, gross income, gross receipts, profits, capital stock, franchise, withholding, payroll, social security, workers compensation, employment, unemployment, disability, property, ad valorem, stamp, excise, severance, occupation, service, sales, use, license, lease, transfer, import, export, value added, alternative minimum, estimated or other similar tax (including any fee, assessment, or other charge in the nature of or in lieu of any tax) imposed by any Tax Authority and any interest, penalties, additions to tax, or additional amounts in respect of the foregoing.

"Tax Authority" means, with respect to any Tax, the governmental entity or political subdivision, agency, commission or authority thereof that imposes such Tax, and the agency, commission or authority (if any) charged with the assessment, determination or collection of such Tax for such entity or subdivision.

"Tax Contest" means an audit, review, examination, or any other administrative or judicial proceeding with the purpose, potential or effect of redetermining Taxes of any member of either Group (including any administrative or judicial review of any claim for refund).

"Tax Counsel" means Baker Botts LLP.

"Tax Item" means, with respect to any Tax, any item of income, gain, loss, deduction, credit or other attribute that may have the effect of increasing or decreasing any Tax.

"Tax Law" means the law of any governmental entity or political subdivision thereof, and any controlling judicial or administrative interpretations of such law, relating to any Tax.

"Tax Materials" means (i) the Ruling and each Supplemental Ruling issued by the IRS in connection with the Distribution, (ii) each IRS Submission and Supplemental IRS Submission, (iii) the representation letters delivered to Tax Counsel in connection with the delivery of the Tax Opinion, and (iv) any other materials delivered or deliverable by Distributing, Spinco and others in connection with the rendering by Tax Counsel of the Tax Opinion or the issuance by the IRS of any Ruling or Supplemental Ruling.

"Tax Opinion" means the opinion to be delivered by Tax Counsel to Distributing in connection with the Distribution to the effect that the Contribution and the Distribution will qualify as a tax-free transaction described under Sections 368(a) and 355 of the Code to Distributing and the holders of Liberty Ventures Common Stock.

"Tax Records" means Tax Returns, Tax Return work papers, documentation relating to any Tax Contests, and any other books of account or records required to be maintained under applicable Tax Laws (including but not limited to Section 6001 of the Code) or under any record retention agreement with any Tax Authority.

"Tax Return" means any report of Taxes due, any claims for refund of Taxes paid, any information return with respect to Taxes, or any other similar report, statement, declaration, or document filed or required to be filed (by paper, electronically or otherwise) under any applicable Tax Law, including any attachments, exhibits, or other materials submitted with any of the foregoing, and including any amendments or supplements to any of the foregoing.

"Tax Year" means, with respect to any Tax, the year, or shorter period, if applicable, for which the Tax is reported as provided under applicable Tax Law.

"Transaction Taxes" means any Taxes resulting from the Restructuring and the Distribution, other than Transfer Taxes.

"Transaction Tax-Related Losses" means any Losses resulting from the failure of (i) the Restructuring to qualify in whole for nonrecognition of income, gain and loss for U.S. federal income tax purposes to Distributing, Spinco and each of their respective Subsidiaries immediately prior to the Distribution, (ii) the Contribution and Distribution to qualify as a tax-free transaction described under Sections 368(a), 355 and 361 of the Code, or (iii) the Contribution and Distribution to qualify in whole for nonrecognition of income, gain and loss for U.S. federal income tax purposes to Distributing, Spinco, each of their respective Subsidiaries at the Effective Time, and the holders of Liberty Ventures Common Stock that receive stock of Spinco in the Distribution.

"Transfer Taxes" means all U.S. federal, state, local or foreign sales, use, privilege, transfer, documentary, gains, stamp, duties, recording, and similar Taxes and fees (including any penalties, interest or additions thereto) imposed upon any party hereto or any of its Subsidiaries in connection with the Restructuring or the Distribution.

"Treasury Regulations" means the regulations promulgated from time to time under the Code as in effect for the relevant Tax Year.

"TripAdvisor" means TripAdvisor, Inc., a Delaware corporation.

SECTION 2. Allocation of Taxes and Tax-Related Losses

2.1 Allocation of Taxes

. Except as provided in Section 2.2 (Special Rules) and Section 3.4(d) (Compensatory Equity Interests), Taxes shall be allocated as follows:

(a)Combined Returns

(i) Allocation of Taxes for Combined Returns

. Distributing shall be allocated: (A) all Taxes that are attributable to members of the

Distributing Group and reported on, or required to be reported on, a Combined Return; and (B) all Taxes that are attributable to members of the Spinco Group for the Pre-Distribution Period and reported on, or required to be reported on, a Combined Return. Spinco shall be allocated all Taxes that are attributable to members of the Spinco Group for the Post-Distribution Period and reported on, or required to be reported on, a Combined Return.

(ii) Transactions Occurring on the Distribution Date

. Notwithstanding the provisions of Section 2.1(a)(i) (but subject to the provisions of Section 2.2), Taxes attributable to any transaction or action taken by or with respect to any member of the Spinco Group outside the ordinary course of business before the Distribution on the Distribution Date shall be allocated to the Pre-Distribution Period, and Taxes attributable to any transaction or action taken by or with respect to any member of the Spinco Group outside the ordinary course of business after the Distribution on the Distribution Date shall be allocated to the Post-Distribution Period.

(b)Separate Returns

.

(i)Spinco Separate Returns

. Spinco shall be allocated all Taxes that are attributable to members of the Spinco Group and reported on, or required to be reported on, a Separate Return that is required to be filed by a member of the Spinco Group.

(ii)Distributing Separate Returns

. Distributing shall be allocated all Taxes that are attributable to members of the Distributing Group and reported on, or required to be reported on, a Separate Return that is required to be filed by a member of the Distributing Group.

(c)Taxes Not Reported on Tax Returns

. Spinco shall be allocated any Tax attributable to members of the Spinco Group that is not required to be reported on a Tax Return, and Distributing shall be allocated any Tax attributable to members of the Distributing Group that is not required to be reported on a Tax Return.

2.2Special Rules

.

(a)Transaction Taxes and Transaction Tax-Related Losses

. Notwithstanding any other provision in this Section 2:

(i)Distributing shall be allocated all Transaction Taxes and Transaction Tax-Related Losses other than any Transaction Taxes and Transaction Tax-Related Losses allocated to Spinco pursuant to clause (ii) of this Section 2

.2(a).

(ii)Spinco will be allocated any Transaction Taxes (including corresponding state and local Taxes) and Transaction Tax-Related Losses that (x) result primarily from, individually or in the aggregate, any breach by Spinco of any of its covenants set forth in Section 7

.1 hereof, or (y) result from a Spinco Section 355(e) Event.

(b)Transfer Taxes

. Notwithstanding any other provision in this Section 2, all Transfer Taxes shall be allocated 50% to Spinco and 50% to Distributing.

2.3Tax Payments

. Each Company shall pay the Taxes allocated to it by this Section 2 either to the applicable Tax Authority or to the other Company in accordance with Section 4 and the other applicable provisions of this Agreement.

SECTION 3.Preparation and Filing of Tax Returns

.

3.1Combined Returns

(a)Preparation of Combined Returns

. Distributing shall be responsible for preparing and filing (or causing to be prepared and filed) all Combined Returns for any Tax Year.

3.2Separate Returns

.

(a) Tax Returns to be Prepared by Distributing

. Distributing shall be responsible for preparing and filing (or causing to be prepared and filed) all Separate Returns which relate to one or more members of the Distributing Group for any Tax Year.

(b)Tax Returns to be Prepared by Spinco

. Spinco shall be responsible for preparing and filing (or causing to be prepared and filed) all Separate Returns which relate to one or more members of the Spinco Group for any Tax Year.

3.3Provision of Information

.

(a)Distributing shall provide to Spinco, and Spinco shall provide to Distributing, any information about members of the Distributing Group or the Spinco Group, respectively, that the Preparer needs to determine the amount of Taxes due on any Payment Date with respect to a Tax Return for which the Preparer is responsible pursuant to Section 3

.1 or 3.2 and to properly and timely file all such Tax Returns.

(b)If a member of the Spinco Group supplies information to a member of the Distributing Group, or a member of the Distributing Group supplies information to a member of the Spinco Group, and an officer of the requesting member intends to sign a statement or other document under penalties of perjury in reliance upon the accuracy of such information, then a duly authorized officer of the member supplying such information shall certify, to the best of such officer's knowledge, the accuracy of the information so supplied

3.4Special Rules Relating to the Preparation of Tax Returns

(a)In General

All Tax Returns that include any members of the Spinco Group or Distributing Group, or any of their respective Affiliates, shall be prepared in a manner that is consistent with the Ruling Request, the Ruling, and the Tax Opinion. Except as otherwise set forth in this Agreement, and subject to Sections 3.4(b) through (d), the Company responsible for preparing and filing (or causing to be prepared and filed) a Tax Return pursuant to Sections 3.1 or 3.2 shall have the right with respect to such Tax Return to determine (i) the manner in which such Tax Return shall be prepared and filed, including the elections, methods of accounting, positions, conventions and principles of taxation to be used and the manner in which any Tax Item shall be reported, (ii) whether any extensions may be requested, (iii) whether an amended Tax Return shall be filed, (iv) whether any claims for refund shall be made, (v) whether any refunds shall be paid by way of refund or credited against any liability for the related Tax and (vi) whether to retain outside firms to prepare or review such Tax Return.

(b)Spinco Tax Returns

. With respect to any Separate Return for which Spinco is responsible pursuant to Section 3.2(b), Spinco and the other members of the Spinco Group must allocate Tax Items between such Separate Return for which Spinco is responsible pursuant to Section 3.2(b) and any related Combined Return for which Distributing is responsible pursuant to Section 3.1 that are filed with respect to the same Tax Year in a manner that is consistent with the reporting of such Tax Items on the related Combined Return for which Distributing is responsible pursuant to Section 3.1.

(c)Election to File Consolidated, Combined or Unitary Tax Returns

. Distributing shall have the sole discretion of filing any Tax Return on a consolidated, combined or unitary basis, if such Tax Return would include at least one member of each Group and the filing of such Tax Return is elective under the relevant Tax Law.

(d)Compensatory Equity Interests

(i) Deductions Related to Compensatory Equity Interests

. To the extent permitted by applicable Tax Law, Income Tax deductions with respect to the issuance, exercise, vesting or settlement after the Distribution Date of any Compensatory Equity Interests held by any Person shall be claimed (A) in the case of an active officer or employee, solely by the Group that employs such Person at the time of such issuance, exercise, vesting, or settlement, as applicable; (B) in the case of a former officer or employee, solely by the Group that was the last to employ such Person; and (C) in the case of a director or former director (who is not an officer or employee or former officer or employee of a member of either Group), (x) solely by the Distributing Group if such person was, at any time before or after the Distribution, a director of any member of the Distributing Group, and (y) in any other case, solely by the Spinco Group (the party whose Group is described in (A), (B), or (C), the "Employing Party").

(ii) Withholding and Reporting

. For any Tax Year (or portion thereof), the Employing Party shall (A) satisfy, or shall cause to be satisfied, all applicable Tax reporting obligations with respect to the issuance, exercise, vesting or settlement of Compensatory Equity Interests and (B) satisfy, or cause to be satisfied, all liabilities for Taxes imposed in connection with such issuance, exercise, vesting or settlement (including the employer portion of any employment taxes); provided that, (x) in the event Compensatory Equity Interests are settled by the corporation that is the issuer or obligor under the Compensatory Equity Interest (the "issuing corporation") and the issuing corporation is not a member of the same Group as the Employing Party, the issuing corporation shall promptly remit to the Employing Party an amount of cash equal to the amount required to be withheld in respect of any withholding Taxes, and (y) the Employing Party shall not be liable for failure to remit to the applicable Tax Authority any amount required to have been withheld from the recipient of the Compensatory Equity Interest in connection with such issuance, exercise, vesting or settlement, except to the extent that the issuing corporation shall have remitted such amount to the Employing Party. Distributing shall promptly notify Spinco, and Spinco shall promptly notify Distributing, regarding the exercise of any option or the issuance, vesting, exercise or settlement of any other Compensatory Equity Interest to the extent that, as a result of such issuance, exercise, vesting or settlement, any other party may be entitled to a deduction or required to pay any Tax, or such information otherwise may be relevant to the preparation of any Tax Return or payment of any Tax by such other party or parties.

(iii)Distributing Employees

. For purposes of this Section 3.4(d), (x) except as described in clause (y) of this Section 3.4(d)(iii), if a Person is an officer or employee of Distributing or any member of the Distributing Group for any Tax Year (or portion thereof), then such officer or employee will exclusively be considered to be employed by Distributing (or the applicable member of the Distributing Group) for such Tax Year (or portion thereof); and (y) if an officer or employee is on the payroll of Buyseasons, Inc. for any Tax Year (or portion thereof), then such officer or employee will exclusively be considered to be employed by Spinco (or the applicable member of the Spinco Group) for such Tax Year (or portion thereof).

(a)Except as otherwise contemplated by this Section 3

.5 or Section 3.6, any refunds, credits or offsets with respect to Taxes of any member of (i) the Distributing Group that were reported on any Combined Return shall be for the account of Distributing, (ii) the Spinco Group that were reported on any Combined Return and are attributable to the Pre-Distribution Period shall be for the account of Distributing, (iii) the Spinco Group that were reported on any Combined Return and are attributable to the Post-Distribution Period shall be for the account of Spinco, (iv) the Distributing Group that were reported on any Separate Return required to be filed by a member of the Distributing Group shall be for the account of Distributing, and (v) the Spinco Group that were reported on any Separate Return required to be filed by a member of the Spinco Group shall be for the account of Spinco.

(b) Notwithstanding Section 3

.5(a), (i) any refunds, credits or offsets with respect to Taxes, including Transaction Taxes, allocated to, and actually paid by, Distributing pursuant to this Agreement shall be for the account of Distributing, and (ii) any refunds, credits or offsets with respect to Taxes, including Transaction Taxes, allocated to, and actually paid by, Spinco pursuant to this Agreement shall be for the account of Spinco.

(c)Distributing shall forward to Spinco, or reimburse Spinco for, any such refunds, credits or offsets, plus any interest received thereon, net of any Taxes incurred with respect to the receipt or accrual thereof and any expenses incurred in connection therewith, that are for the account of Spinco within five business days from receipt thereof by Distributing or any of its Affiliates

. Spinco shall forward to Distributing, or reimburse Distributing for, any refunds, credits or offsets, plus any interest received thereon, net of any Taxes incurred with respect to the receipt or accrual thereof and any expenses incurred in connection therewith, that are for the account of Distributing within five business days from receipt thereof by Spinco or any of its Affiliates. Any refunds, credits or offsets, plus any interest received thereon, or reimbursements not forwarded or made within the five business day period specified above shall bear interest from the date received by the refunding or reimbursing party (or its Affiliates) through and including the date of payment at the Interest Rate (treating the date received as the Due Date for purposes of determining such Interest Rate). If, subsequent to a Tax Authority's allowance of a refund, credit or offset, such Tax Authority reduces or eliminates such allowance, any refund, credit or offset, plus any interest received thereon, forwarded or reimbursed under this Section 3.5 shall be returned to the party who had forwarded or reimbursed such refund, credit or offset and interest upon the request of such forwarding party in an amount equal to the applicable reduction, including any interest received thereon.

3.6Carrybacks

. If and to the extent that Spinco requests in writing that Distributing or any of its Affiliates obtain a refund, credit or offset of Taxes with respect to the carryback of any Tax attribute of the members of the Spinco Group arising in a Post-Distribution Period to a Pre-Distribution Period, and provided that Distributing or any of its Affiliates would not otherwise be required to forego a refund, credit or offset of Taxes for its own account or otherwise be adversely affected as a result of such carryback, then

(i) Distributing (or its Affiliate) shall take all reasonable measures to obtain a refund, credit or offset of Tax with respect to such carryback (including by filing an amended Tax Return), and (ii) to the extent that Distributing or any of its Affiliates receives any refund, credit or offset of Taxes attributable (on a last dollar basis) to such carryback, Distributing shall pay such refund, credit or offset, plus any interest received thereon, to Spinco within five business days from receipt thereof by Distributing or any of its Affiliates; *provided, however*, that Distributing shall be entitled to reduce the amount of any such refund, credit or offset for its reasonable out-of-pocket costs and expenses incurred in connection therewith and any Taxes incurred with respect to the receipt or accrual thereof; and *provided further*, that Spinco, upon the request of Distributing, agrees to repay such refund, credit or offset, plus any interest received thereon and net of Taxes, to Distributing in the event, and to the extent, that Distributing is required to repay such refund, credit or offset, plus any interest received thereon, to a Tax Authority.

3.7Amended Returns

. Any amended Tax Return or claim for Tax refund, credit or offset with respect to any member of the Spinco Group may be made only by the Company (or its Subsidiaries) responsible for preparing the original Tax Return with respect to such member pursuant to Sections 3.1 and 3.2. Spinco (or its Subsidiaries) shall not, without the prior written consent of Distributing (which consent shall not be unreasonably withheld or delayed), file, or cause to be filed, any such amended Tax Return or claim for Tax refund, credit or offset to the extent that such filing, if accepted, is likely to increase the Taxes allocated to, or the Tax indemnity obligations under this Agreement of, Distributing for any Tax Year (or portion thereof) by more than a *de minimis* amount; *provided, however*, that such consent need not be obtained if Spinco agrees to indemnify Distributing for the incremental Taxes allocated to, or the incremental Tax indemnity obligation resulting under this Agreement to, Distributing as a result of the filing of such amended Tax Return.

SECTION 4.Tax Payments

٠

4.1Payment of Taxes to Tax Authority

. Distributing shall be responsible for remitting to the proper Tax Authority the Tax shown on any Tax Return for which it is responsible for the preparation and filing pursuant to Section 3.1 or Section 3.2, and Spinco shall be responsible for remitting to the proper Tax Authority the Tax shown on any Tax Return for which it is responsible for the preparation and filing pursuant to Section 3.2.

4.2Indemnification Payments

.

(a) Tax Payments Made by the Distributing Group

. If any member of the Distributing Group is required to make a payment to a Tax Authority for Taxes allocated to Spinco under this Agreement, Spinco will pay the amount of Taxes allocated to it to Distributing not later than the later of (i) five business days after receiving notification requesting such amount, and (ii) one business day prior to the date such payment is required to be made to such Tax Authority.

(b) Tax Payments Made by the Spinco Group

. If any member of the Spinco Group is required to make a payment to a Tax Authority for Taxes allocated to Distributing under this Agreement, Distributing will pay the amount of Taxes allocated to it to Spinco not later than the later of (i) five business days after receiving notification requesting such amount, and (ii) one business day prior to the date such payment is required to be made to such Tax Authority.

4.3Interest on Late Payments

. Payments pursuant to this Agreement that are not made by the date prescribed in this Agreement or, if no such date is prescribed, not later than five business days after demand for payment is made (the "Due Date") shall bear interest for the period from and including the date immediately following the Due Date through and including the date of payment at the Interest Rate. Such interest will be payable at the same time as the payment to which it relates.

4.4Tax Consequences of Payments

For all Tax purposes and to the extent permitted by applicable Tax Law, the parties hereto shall treat any payment made pursuant to this Agreement as a capital contribution or a distribution, as the case may be, immediately prior to the Distribution. If the receipt or accrual of any indemnity payment under this Agreement causes, directly or indirectly, an increase in the taxable income of the recipient under one or more applicable Tax Laws, such payment shall be increased so that, after the payment of any Taxes with respect to the payment, the recipient thereof shall have realized the same net amount it would have realized had the payment not resulted in taxable income. To the extent that Taxes for which any party hereto (the indemnifying party) is required to pay another party (the indemnified party) pursuant to this Agreement may be deducted or credited in determining the amount of any other Taxes required to be paid by the indemnified party (for example, state Taxes which are permitted to be deducted in determining federal Taxes), the amount of any payment made to the indemnified party by the indemnifying party shall be decreased by taking into account any resulting reduction in other Taxes of the indemnified party. If such a reduction in Taxes of the indemnified party occurs following the payment made to the indemnified Party with respect to the relevant indemnified Taxes, the indemnified party shall promptly repay the indemnifying party the amount of such reduction when actually realized. If the Tax benefit arising from the foregoing reduction of Taxes described in this Section 4.4 is subsequently decreased or eliminated, then the indemnifying party shall promptly pay the indemnified party the amount of the decrease in such Tax benefit.

SECTION 5. Assistance and Cooperation

5.1Cooperation

. In addition to the obligations enumerated in Sections 3.3 and 7.7, Distributing and Spinco will cooperate (and cause their respective Subsidiaries and Affiliates to cooperate) with each other and with each other's agents, including accounting firms and legal counsel, in connection with Tax matters, including provision of relevant documents and information in their possession and making available to each other, as reasonably requested and available, personnel (including officers, directors, employees and agents of the parties or their respective Subsidiaries or Affiliates) responsible for preparing, maintaining, and interpreting information and documents relevant to Taxes, and

personnel reasonably required as witnesses or for purposes of providing information or documents in connection with any administrative or judicial proceedings relating to Taxes.

5.2Supplemental Rulings

(a)Each of the parties agrees that, at the reasonable request of another party (the "Requesting Party"), Distributing and Spinco shall (and shall cause each member of its Group) to cooperate and use reasonable best efforts to obtain, as expeditiously as reasonably practicable, a Supplemental Ruling from the IRS

. Notwithstanding the foregoing, Distributing shall not be required to file any Supplemental IRS Submission unless Spinco represents to Distributing that (i) it has reviewed the Supplemental IRS Submission, and (ii) all information and representations, if any, relating to any member of the Spinco Group and their Affiliates contained in the Supplemental IRS Submission are true, correct and complete in all material respects. The Requesting Party shall reimburse the other parties for all reasonable out-of-pocket costs and expenses incurred by such parties or their Affiliates in connection with obtaining or requesting such Supplemental Ruling within five business days after receiving an invoice from such party therefor.

(b)Distributing shall provide Spinco with a reasonable opportunity to review and comment on each Supplemental IRS Submission to be filed by Distributing prior to the filing of such Supplemental IRS Submission with the IRS, and Spinco shall provide Distributing with a reasonable opportunity to review and comment on each Supplemental IRS Submission to be filed by Spinco prior to the filing of such Supplemental IRS Submission with the IRS

. No Supplemental IRS Submission shall be filed by Spinco with the IRS unless, prior to such filing Distributing shall have agreed as to the contents of such Supplemental IRS Submission to the extent that the Supplemental IRS Submission (i) includes statements or representations relating to facts that are or will be under the control of any member of the Distributing Group or any of its Affiliates or (ii) is relevant to, or creates, any actual or potential obligations of, or limitations on, any member of the Distributing Group or any of their Affiliates; provided, however, that if the IRS requests same-day filing of a Supplemental IRS Submission that does not include any material issue or statement, then Spinco is required only to make a good faith effort to notify Distributing's representatives and to give such representatives an opportunity to review and comment on such Supplemental IRS Submission prior to filing it with the IRS. No Supplemental IRS Submission shall be filed by Distributing with the IRS unless, prior to the filing, Spinco shall have agreed as to the contents of such Supplemental IRS Submission to the extent that the Supplemental IRS Submission (i) includes statements or representations relating to facts that are or will be under the control of any member of the Spinco Group or any of its Affiliates or (ii) is relevant to, or creates, any actual or potential obligations of, or limitations on, any member of the Spinco Group or any of their Affiliates; provided, however, that if the IRS requests sameday filing of a Supplemental IRS Submission that does not include any material issue or statement, then Distributing is required only to make a good faith effort to notify Spinco's representatives and to give such representatives an opportunity to review and comment on such Supplemental IRS Submission prior to filing it with the IRS. Each Company (or its representatives) shall provide the other Company (or its representatives) with copies of each Supplemental IRS

Submission filed with the IRS promptly following the filing thereof. Neither Company nor its representatives shall conduct any substantive communications with the IRS regarding any material issue arising with respect to any Supplemental Ruling, including meetings or conferences with IRS personnel, whether telephonically, in person or otherwise, without first notifying the other Company (or its representatives) and giving the other Company (or its representatives) a reasonable opportunity to participate, and a reasonable number of such Company's representatives shall have an opportunity to participate in all conferences or meetings with IRS personnel that take place in person, regardless of the nature of the issues expected to be discussed; provided, however, that in the case of communications concerning a Supplemental Ruling that occur during an unscheduled conference initiated by the IRS or a conference initiated by a Company or its representatives for a purpose unrelated to a Supplemental Ruling in connection with which it is not reasonably practicable to provide to the other Company or its representatives advance notice and an opportunity to participate, such Company (or its representatives) shall promptly update the other Company and its representatives as to the content of such communications. Each Company shall promptly provide the other Company (or its representatives) with copies of any correspondence received by such Company (or its representatives) from the IRS relating to any Supplemental Ruling.

SECTION 6.Tax Records

•

6.1Retention of Tax Records

. Each of Distributing and Spinco shall preserve, and shall cause their respective Subsidiaries to preserve, all Tax Records that are in their possession, and that could affect the liability of any member of the other Group for Taxes, for as long as the contents thereof may become material in the administration of any matter under applicable Tax Law, but in any event until the later of (x) the expiration of any applicable statutes of limitation, as extended, and (y) seven years after the Distribution Date.

6.2Access to Tax Records

Spinco shall make available, and cause its Subsidiaries to make available, to members of the Distributing Group for inspection and copying (x) all Tax Records in their possession that relate to a Pre-Distribution Period, and (y) the portion of any Tax Record in their possession that relates to a Post-Distribution Period and which is reasonably necessary for the preparation of a Tax Return by a member of the Distributing Group or any of their Affiliates or with respect to a Tax Contest by a Tax Authority of such return. Distributing shall make available, and cause its Subsidiaries to make available, to members of the Spinco Group for inspection and copying the portion of any Tax Record in their possession that relates to a Pre-Distribution Period and which is reasonably necessary for the preparation of a Tax Return by a member of the Spinco Group or any of their Affiliates or with respect to a Tax Contest by a Tax Authority of such return.

6.3Confidentiality

. Each party hereby agrees that it will hold, and shall use its reasonable best efforts to cause its officers, directors, employees, accountants, counsel, consultants, advisors and agents to hold, in confidence all records and information prepared and shared by and among the parties in carrying out the intent of this Agreement, except as may otherwise be necessary in connection with the filing of Tax Returns or any

administrative or judicial proceedings relating to Taxes or unless disclosure is compelled by a governmental authority. Information and documents of one party (the "Disclosing Party") shall not be deemed to be confidential for purposes of this Section 6.3 to the extent such information or document (i) is previously known to or in the possession of the other party or parties (the "Receiving Party") and is not otherwise subject to a requirement to be kept confidential, (ii) becomes publicly available by means other than unauthorized disclosure under this Agreement by the Receiving Party or (iii) is received from a third party without, to the knowledge of the Receiving Party after reasonable diligence, a duty of confidentiality owed to the Disclosing Party.

6.4Delivery of Tax Records

. Within five business days after receiving notification from Spinco requesting any applicable Tax Records described below which are in the possession of a member of the Distributing Group, Distributing shall provide to Spinco (to the extent not previously provided or held by any member of the Spinco Group on the Distribution Date) copies of (i) the Separate Returns of any member of the Spinco Group, (ii) the relevant portions of any other Tax Returns with respect to any member of the Spinco Group, and (iii) other existing Tax Records (or the relevant portions thereof) reasonably necessary to prepare and file any Tax Returns of, or with respect to, the members of the Spinco Group, or to defend or contest Tax matters relevant to the members of the Spinco Group, including in each case, all Tax Records related to Tax attributes of the members of the Spinco Group and any and all communications or agreements with, or rulings by, any Tax Authority with respect to any member of the Spinco Group.

SECTION 7.Restriction on Certain Actions of Distributing and Spinco; Indemnity

7.1Restrictive Covenants

(a)General Restrictions

. Following the Effective Time, Spinco shall not, and shall cause the members of the Spinco Group and their Affiliates not to, and Distributing shall not, and shall cause the members of the Distributing Group and their Affiliates not to, take any action that, or fail to take any action the failure of which, (i) would cause Distributing or any Subsidiary of Distributing immediately prior to the Distribution to recognize gain or loss, or otherwise include any amount in income, as a result of the Restructuring for U.S. federal income tax purposes, (ii) would be inconsistent with the Contribution and Distribution qualifying, or would preclude the Contribution and Distribution from qualifying, as a tax-free transaction described under Sections 368(a), 355 and 361 of the Code, or (iii) would cause Distributing, Spinco, any of their respective Subsidiaries at the Effective Time, or the holders of Liberty Ventures Common Stock that receive stock of Spinco in the Distribution, to recognize gain or loss, or otherwise include any amount in income, as a result of the Contribution and/or the Distribution for U.S. federal income tax purposes.

(b)Restricted Actions

. Without limiting the provisions of Section 7.1(a) hereof, following the Effective Time, Spinco shall not, and shall cause the members of the Spinco Group and their Affiliates not to, and Distributing shall not, and shall cause the members of the Distributing Group and their Affiliates not to, take any

action that, or fail to take any action the failure of which, would be inconsistent with, or would cause any Person to be in breach of, any representation or covenant, or any material statement, made in the Tax Materials.

(c)Reporting

. Unless and until there has been a Final Determination to the contrary, each party agrees not to take any position on any Tax Return, in connection with any Tax Contest, or otherwise for Tax purposes (in each case, excluding any position taken for financial accounting purposes) that is inconsistent with the Ruling or the Tax Opinion.

7.2Distributing Indemnity

. Distributing agrees to indemnify and hold harmless each member of the Spinco Group and their respective directors, officers, employees, agents, successors and assigns (the "Spinco Indemnitees") from and against any and all (without duplication) (a) Taxes allocated to Distributing pursuant to Section 2.1, (b) Transaction Taxes and Transaction Tax-Related Losses allocated to Distributing pursuant to Section 2.2, (c) Taxes and Losses arising out of or based upon any breach or nonperformance of any covenant or agreement made or to be performed by Distributing contained in this Agreement, (d) Transfer Taxes allocated to Distributing pursuant to Section 2.2, and (e) reasonable out-of-pocket legal, accounting and other advisory and court fees and expenses incurred in connection with the items described in clauses (a) through (d); provided, however, that notwithstanding clauses (a), (c) and (e) of this Section 7.2, Distributing shall not be responsible for, and shall have no obligation to indemnify or hold harmless any Spinco Indemnitee for, (x) any Transaction Taxes or Transaction Tax-Related Losses that are allocated to Spinco pursuant to Section 2.2, or (y) any Taxes or Losses arising out of or based upon any breach or nonperformance of any covenant or agreement made or to be performed by Spinco contained in this Agreement.

7.3 Spinco Indemnity

. Spinco agrees to indemnify and hold harmless each member of the Distributing Group and their respective directors, officers, employees, agents, successors and assigns (the "Distributing Indemnitees") from and against any and all (without duplication) (a) Taxes allocated to Spinco pursuant to Section 2.1, (b) Transaction Taxes and Transaction Tax-Related Losses allocated to Spinco pursuant to Section 2.2, (c) Taxes and Losses arising out of or based upon any breach or nonperformance of any covenant or agreement made or to be performed by Spinco contained in this Agreement, (d) Transfer Taxes allocated to Spinco pursuant to Section 2.2, and (e) reasonable out-of-pocket legal, accounting and other advisory and court fees and expenses incurred in connection with the items described in clauses (a) through (d); provided, however, that notwithstanding clauses (a), (c) and (e) of this Section 7.3, Spinco shall not be responsible for, and shall have no obligation to indemnify or hold harmless any Distributing Indemnitee for, (x) any Transaction Taxes or Transaction Tax-Related Losses that are allocated to Distributing pursuant to Section 2.2, or (y) any Taxes or Losses arising out of or based upon any breach or nonperformance of any covenant or agreement made or to be performed by Distributing contained in this Agreement.

7.4Scope

. The provisions of this Section 7 are intended to be for the benefit of, and shall be enforceable by, each Distributing Indemnitee and its successors in interest and each Spinco Indemnitee and its successors in interest.

7.5 Notices of Tax Contests (Other than Joint Claims)

. Each Company shall provide prompt notice to the other Company of any pending or threatened Tax audit, assessment or proceeding or other Tax Contest of which it becomes aware relating to Taxes for which it is or may be indemnified by such other Company hereunder (other than any Transaction Taxes which shall be governed by Section 7.8). Such notice shall contain factual information (to the extent known) describing any asserted Tax liability in reasonable detail and shall be accompanied by copies of any notice and other documents received from any Tax Authority in respect of any such matters; provided, however, that failure to give such notification shall not affect the indemnification provided hereunder except, and only to the extent that, the indemnifying Company shall have been actually prejudiced as a result of such failure. Thereafter, the indemnified Company shall deliver to the indemnifying Company such additional information with respect to such Tax Contest in its possession that the indemnifying Company may reasonably request.

7.6Control of Tax Contests (Other than Joint Claims)

(a)General Rule

. Except as provided in Sections 7.6(b) and 7.8, each Company (or the appropriate member of its Group) shall have full responsibility, control and discretion in handling, defending, settling or contesting any Tax Contest involving a Tax reported (or that, it is asserted, should have been reported) on a Tax Return for which such Company is responsible for preparing and filing (or causing to be prepared and filed) pursuant to Section 3 of this Agreement.

(b) Non-Preparer Participation Rights

. With respect to a Tax Contest (other than with respect to a Joint Claim) of any Tax Return which could result in a Tax liability for which the Non-Preparer may be liable under this Agreement, (i) the Non-Preparer shall, at its own cost and expense, be entitled to participate in such Tax Contest, (ii) the Preparer shall keep the Non-Preparer updated and informed, and shall consult with the Non-Preparer, (iii) the Preparer shall act in good faith with a view to the merits in connection with the Tax Contest, and (iv) the Preparer shall not settle or compromise such Tax Contest without the prior written consent of the Non-Preparer (which consent shall not be unreasonably withheld or delayed) if the settlement or compromise could have a more than de minimis impact on the Non-Preparer or its Affiliates.

7.7Cooperation

. The parties shall provide each other with all information relating to a Tax Contest which is needed by the other party or parties to handle, participate in, defend, settle or contest the Tax Contest. At the request of any party, the other parties shall take any action (e.g., executing a power of attorney) that is reasonably necessary in order for the requesting party to exercise its rights under this Agreement in respect of a Tax Contest. Spinco shall assist Distributing, and Distributing shall assist Spinco, in taking any remedial actions that are necessary or desirable to minimize the effects of any adjustment made by a Tax Authority. The indemnifying party or parties shall reimburse the indemnified party or parties for any reasonable out-of-pocket costs and expenses incurred in complying with this Section 7.7.

7.8Joint Claims

. Each Company shall promptly give notice to the other Company of any pending or threatened Tax Contest, claim, action, suit, investigation or

proceeding brought by a third party relating to any Transaction Taxes or Transaction Tax-Related Losses for which such Company is or may be indemnified by the other Company under this Section 7 (each, a "Joint Claim"). Such notice shall contain (i) factual information (to the extent known) describing any asserted Tax liability or other claim in reasonable detail and shall be accompanied by copies of any notice and other documents received from any Tax Authority or third party relating to the Joint Claim, and (ii) the amount of the Joint Claim. Such notice shall be given within a reasonable period of time after notice thereof was received by such Company, but any failure to give timely notice shall not affect the indemnities given hereunder except, and only to the extent that, the indemnifying Company shall have been actually prejudiced as a result of such failure. Thereafter, each Company shall deliver to the other Company such additional information with respect to such Joint Claim in its possession that the other Company may reasonably request. Distributing and Spinco will have the right to jointly control the defense, compromise or settlement of any Joint Claim. No indemnified Company shall settle or compromise or consent to entry of any judgment with respect to any such Joint Claim without the prior written consent of the indemnifying Company, which consent may be withheld in the indemnifying Company's sole discretion. No indemnifying Company shall settle or compromise or consent to entry of any judgment with respect to any such Joint Claim without the prior written consent of the indemnified Company, which consent may not be unreasonably withheld or delayed.

SECTION 8.General Provisions

.

8.1Termination

. This Agreement shall terminate at such time as all obligations and liabilities of the parties hereto have been satisfied. The obligations and liabilities of the parties arising under this Agreement shall continue in full force and effect until all such obligations have been satisfied and such liabilities have been paid in full, whether by expiration of time, operation of law, or otherwise.

8.2Predecessors or Successors

. Any reference to Distributing, Spinco, TripAdvisor, a Person, or a Subsidiary in this Agreement shall include any predecessors or successors (*e.g.*, by merger or other reorganization, liquidation, conversion, or election under Treasury Regulations Section 301.7701-3) of Distributing, Spinco, TripAdvisor, such Person, or such Subsidiary, respectively.

8.3Expenses

. Except as otherwise expressly provided for herein, each party and its Subsidiaries shall bear their own expenses incurred in connection with preparation of Tax Returns and other matters related to Taxes under the provisions of this Agreement for which they are liable.

8.4Governing Law

. THIS AGREEMENT SHALL BE GOVERNED BY AND CONSTRUED IN ACCORDANCE WITH THE LAWS OF THE STATE OF DELAWARE, WITHOUT GIVING EFFECT TO THE PRINCIPLES OF CONFLICTS OF LAW THEREOF. Any suit, action or proceeding seeking to enforce any provision of, or based on any matter arising out of or in connection with, this Agreement will be brought exclusively in the Court of Chancery of the State of Delaware (the "Delaware Chancery Court"), or, if the Delaware Chancery Court does not have subject matter jurisdiction, in

the federal courts located in the State of Delaware. Each of the parties hereby consents to personal jurisdiction in any such action, suit or proceeding brought in any such court (and of the appropriate appellate courts therefrom) and irrevocably waives, to the fullest extent permitted by law, any objection that it may now or hereafter have to the laying of the venue of any such suit, action or proceeding in any such court or that any such suit, action or proceeding brought in any such court has been brought in an inconvenient forum. Process in any such suit, action or proceeding may be served on any party anywhere in the world, whether within or without the jurisdiction of any such court. Without limiting the foregoing, each party agrees that service of process on such party as provided in Section 8.6 shall be deemed effective service of process on such party.

8.5Waiver of Jury Trial

EACH PARTY HERETO ACKNOWLEDGES AND AGREES THAT ANY CONTROVERSY WHICH MAY ARISE UNDER THIS AGREEMENT IS LIKELY TO INVOLVE COMPLICATED AND DIFFICULT ISSUES, AND, THEREFORE, EACH SUCH PARTY HEREBY IRREVOCABLY AND UNCONDITIONALLY WAIVES TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, ANY RIGHT SUCH PARTY MAY HAVE TO A TRIAL BY JURY IN RESPECT TO ANY ACTION DIRECTLY OR INDIRECTLY ARISING OUT OF, UNDER OR IN CONNECTION WITH OR RELATING TO THIS AGREEMENT. EACH PARTY HERETO CERTIFIES AND ACKNOWLEDGES THAT (A) NO REPRESENTATIVE, AGENT OR ATTORNEY OF ANY OTHER PARTY HERETO HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH OTHER PARTY WOULD NOT, IN THE EVENT OF SUCH ACTION, SEEK TO ENFORCE THE FOREGOING WAIVER, (B) EACH SUCH PARTY UNDERSTANDS AND HAS CONSIDERED THE IMPLICATIONS OF THIS WAIVER, (C) EACH SUCH PARTY MAKES THIS WAIVER VOLUNTARILY, AND (D) EACH SUCH PARTY HAS BEEN INDUCED TO ENTER INTO THIS AGREEMENT BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS AND CERTIFICATIONS IN THIS SECTION 8.5.

8.6Notices

. All notices and other communications hereunder shall be in writing and shall be delivered in person, by facsimile (with confirming copy sent by one of the other delivery methods specified herein), by overnight courier or sent by certified, registered or express air mail, postage prepaid, and shall be deemed given when so delivered in person, or when so received by facsimile or courier, or, if mailed, three (3) calendar days after the date of mailing, as follows:

(a) If to Distributing, to:

Liberty Interactive Corporation 12300 Liberty Boulevard Englewood, Colorado 80112

Attn: Albert Rosenthaler Facsimile:

(b) If to Spinco, to:

Liberty TripAdvisor Holdings, Inc. 12300 Liberty Boulevard Englewood, Colorado 80112

Attn: Tim Lenneman Facsimile:

or to such other address as the party to whom notice is given may have previously furnished to the other parties in writing in the manner set forth above.

8.7Counterparts

. This Agreement may be executed in two or more identical counterparts, each of which shall be deemed to be an original, and all of which together shall constitute one and the same agreement. The Agreement may be delivered by facsimile transmission of a signed copy thereof.

8.8Binding Effect; Assignment

. This Agreement and all of the provisions hereof shall be binding upon and inure to the benefit of the parties hereto and their respective successors and permitted assigns. Except with respect to a merger of a party, neither this Agreement nor any of the rights, interests or obligations hereunder shall be assigned by any party hereto without the prior written consent of the other parties; provided, however, that each of Distributing and Spinco may assign its respective rights, interests, duties, liabilities and obligations under this Agreement to any other member of their Group, but such assignment shall not relieve Distributing or Spinco, as the assignor, of its liabilities or obligations hereunder.

8.9Severability

. Any provision of this Agreement which is prohibited or unenforceable in any jurisdiction shall, as to such jurisdiction, be ineffective to the extent of such prohibition or unenforceability without invalidating the remaining provisions hereof. Any such prohibition or unenforceability in any jurisdiction shall not invalidate or render unenforceable such provision in any other jurisdiction. Upon a determination that any provision of this Agreement is prohibited or unenforceable in any jurisdiction, the parties shall negotiate in good faith to modify this Agreement so as to effect the original intent of the parties as closely as possible in an acceptable manner in order that the provisions contemplated hereby are consummated as originally contemplated to the fullest extent possible.

8.10Amendments; Waivers

. Any provision of this Agreement may be amended or waived if, but only if, such amendment or waiver is in writing and is signed, in the case of an amendment, by each party to this Agreement, or in the case of a waiver, by the party against whom the waiver is to be effective. No failure or delay by any party in exercising any right, power or privilege hereunder shall operate as a waiver thereof nor shall any single or partial exercise thereof preclude any other or further exercise thereof or the exercise of any other right, power or privilege. Except as otherwise provided herein, the rights and remedies herein provided shall be cumulative and not exclusive of any rights

or remedies provided by applicable law. Any consent provided under this Agreement must be in writing, signed by the party against whom enforcement of such consent is sought.

8.11Effective Date

. This Agreement shall become effective on the date recited above on which the parties entered into this Agreement.

8.12Change in Law

. Any reference to a provision of the Code or any other Tax Law shall include a reference to any applicable successor provision or law.

8.13 Authorization, Etc.

Each of

the parties hereto hereby represents and warrants that it has the power and authority to execute, deliver and perform this Agreement, that this Agreement has been duly authorized by all necessary corporate action on the part of such party, that this Agreement constitutes a legal, valid and binding obligation of such party and that the execution, delivery and performance of this Agreement by such party does not contravene or conflict with any provision of law or of its charter or bylaws or any agreement, instrument or order binding such party

8.14No Third Party Beneficiaries

. Except as provided in Sections 7.2 and 7.3 of this Agreement, this Agreement is solely for the benefit of Distributing, Spinco, and their Subsidiaries and is not intended to confer upon any other Person any rights or remedies hereunder. Notwithstanding anything in this Agreement to the contrary, this Agreement is not intended to confer upon any Spinco Indemnitees any rights or remedies against Spinco hereunder, and this Agreement is not intended to confer upon any Distributing Indemnitees any rights or remedies against Distributing hereunder.

8.15Entire Agreement

. This Agreement embodies the entire understanding among the parties relating to its subject matter and supersedes and terminates any prior agreements and understandings among the parties with respect to such subject matter, and no party to this Agreement shall have any right, responsibility, obligation or liability under any such prior agreement or understanding. Any and all prior correspondence, conversations and memoranda are merged herein and shall be without effect hereon. No promises, covenants or representations of any kind, other than those expressly stated herein, have been made to induce any party to enter into this Agreement.

8.16No Strict Construction; Interpretation

.

(a)Distributing and Spinco each acknowledge that this Agreement has been prepared jointly by the parties hereto and shall not be strictly construed against any party hereto

•

(b)When a reference is made in this Agreement to an Article, Section, Exhibit or Schedule, such reference shall be to an Article of, a Section of, or an Exhibit or Schedule to, this Agreement unless otherwise indicated

. The table of contents and headings contained in this Agreement are for reference purposes only and shall not affect in any way the meaning or interpretation of this Agreement. Whenever the words "include", "includes" or "including" are used in this Agreement, they shall be deemed to be followed by the words "without limitation". The words "hereof", "herein" and "hereunder" and words of similar import when used in this Agreement shall refer to this

Agreement as a whole and not to any particular provision of this Agreement. All terms defined in this Agreement shall have the defined meanings when used in any certificate or other document made or delivered pursuant hereto unless otherwise defined therein. The definitions contained in this Agreement are applicable to the singular as well as the plural forms of such terms and to the masculine as well as to the feminine and neuter genders of such term. Any agreement, instrument or statute defined or referred to herein or in any agreement or instrument that is referred to herein means such agreement, instrument or statute as from time to time amended, modified or supplemented, including (in the case of agreements or instruments) by waiver or consent and (in the case of statutes) by succession of comparable successor statutes and references to all attachments thereto and instruments incorporated therein. References to a Person are also to its permitted successors and assigns.

8.17Headings

. The headings contained in this Agreement are for reference purposes only and will not affect in any way the meaning or interpretation of this Agreement.

IN WITNESS WHEREOF, the parties have caused this Agreement to be executed by the respective officers as of the date set forth above.

LIBERTY INTERACTIVE CORPORATION

By: /s/ Albert Rosenthaler

Name: Albert Rosenthaler Title: Senior Vice President

LIBERTY TRIPADVISOR HOLDINGS, INC.

By: /s/ Tim Lenneman

Name: Tim Lenneman Title: Vice President

28



August 27, 2014

Liberty Interactive Corporation Announces Completion of Liberty TripAdvisor Holdings Spin-Off

ENGLEWOOD, Colo.--Liberty Interactive Corporation ("Liberty") (Nasdaq: LINTA, LINTB, LVNTA, LVNTB) announced that it has completed the spin-off (the "Spin-Off") of Liberty TripAdvisor Holdings, Inc. ("TripAdvisor Holdings") (Nasdaq: LTRPA, LTRPB) at 5:00 p.m., New York City time, today. As a result, Liberty and TripAdvisor Holdings are now separate publicly traded companies. TripAdvisor Holdings will begin trading regular way on Thursday, August 28, 2014 on the Nasdaq Stock Market, under the symbols listed above.

About Liberty Interactive Corporation

Liberty Interactive Corporation operates and owns interests in a broad range of digital commerce businesses. Those interests are currently attributed to two tracking stock groups: Liberty Interactive Group and Liberty Ventures Group. The businesses and assets attributed to the Liberty Interactive Group (Nasdaq: LINTA, LINTB) are primarily focused on digital commerce and consist of Liberty Interactive Corporation's subsidiaries QVC, Provide Commerce, Backcountry.com, Bodybuilding.com, Evite, CommerceHub and its interest in HSN. The businesses and assets attributed to the Liberty Ventures Group (Nasdaq: LVNTA, LVNTB) consist of all of Liberty Interactive Corporation's businesses and assets other than those attributed to the Liberty Interactive Group and include its interest in Expedia, and minority interests in Time Warner, Time Warner Cable, Lending Tree and Interval Leisure Group._

About Liberty TripAdvisor Holdings

Liberty TripAdvisor Holdings' businesses consist of its subsidiaries TripAdvisor and BuySeasons. TripAdvisor is the world's largest online travel community, aggregating reviews and opinions from its community of travelers about destinations, accommodations, restaurants and activities throughout the world. BuySeasons is a leading online retailer of costumes and party supplies.

CONTACT: Courtnee Ulrich 720-875-5420

1

QVC to Present at Goldman Sachs 21st Annual Global Retailing Conference

ENGLEWOOD, Colo.--(BUSINESS WIRE)-- Liberty Interactive Corporation (Nasdaq: LINTA, LINTB, LVNTA, LVNTB) announced that Michael George, President and CEO of QVC, Inc., will be presenting at the Goldman Sachs 21st Annual Global Retailing Conference on Thursday, September 4th at 8:50 a.m., E.D.T. at the Plaza Hotel in New York, NY. During his presentation, Mr. George may make observations regarding the company's financial performance and outlook

The presentation will be broadcast live via the Internet. All interested persons should visit the Liberty Interactive Corporation website at http://www.libertyinteractive.com/events to register for the webcast. An archive of the webcast will also be available on this website for 30 days after appropriate filings have been made with the SEC.

About Liberty Interactive Corporation

Liberty Interactive Corporation operates and owns interests in a broad range of digital commerce businesses. Those interests are currently attributed to two tracking stock groups: Liberty Interactive Group and Liberty Ventures Group. The businesses and assets attributed to the Liberty Interactive Group (Nasdaq: LINTA, LINTB) are primarily focused on digital commerce and consist of Liberty Interactive Corporation's subsidiaries QVC, Provide Commerce, Backcountry.com, Bodybuilding.com, Evite, CommerceHub and its interest in HSN. The businesses and assets attributed to the Liberty Ventures Group (Nasdaq: LVNTB, LVNTB) consist of all of Liberty Interactive Corporation's businesses and assets other than those attributed to the Liberty Interactive Group and include its interest in Expedia, and minority interests in Time Warner Cable, Lending Tree and Interval Leisure Group.

Liberty Interactive Corporation

Courtnee Ulrich, 720-875-5420

Source: Liberty Interactive Corporation

Liberty Interactive Corporation Condensed Pro Forma Consolidated Financial Statements

(unaudited)

On August 27, 2014, Liberty Interactive Corporation ("Liberty" or the "Company") completed the distribution to the stockholders of Liberty shares of a wholly-owned subsidiary, Liberty TripAdvisor Holdings, Inc. ("TripCo"), which held the subsidiaries TripAdvisor, Inc. ("TripAdvisor") and BuySeasons, Inc. which includes the retail businesses BuyCostumes.com and Celebrate Express ("BuySeasons") (the "Trip Spin-Off"), both of which operate as standalone operating entities. The transaction was effected as a pro-rata dividend of shares of TripCo to the stockholders of Series A and Series B Liberty Ventures common stock of Liberty.

The following unaudited condensed pro forma consolidated financial statements have been prepared giving effect to the Trip Spin-Off as if it occurred as of December 31, 2013 and March 31, 2014 for the condensed pro forma consolidated balance sheets and January 1, 2013 for the condensed pro forma consolidated statements of operation. The unaudited condensed pro forma consolidated financial statements do not purport to represent what Liberty's financial position actually would have been had the Trip Spin-Off occurred on the dates indicated or to project Liberty's operating results for any future period. The unaudited condensed pro forma consolidated financial statements should be read in conjunction with the publicly available information of Liberty, including the Form 10-K, as filed on February 28, 2014 with the Securities and Exchange Commission (the "SEC").

Liberty Interactive Corporation Condensed Pro Forma Consolidated Balance Sheet December 31, 2013 (unaudited)

	Liberty Intera historical (1		Less: TripCo historical (2)	Liberty Interactive pro forma
<u>Assets</u>				
Cash and cash equivalents	\$	1,256	354	902
Other current assets		3,170	289	2,881
Investments in available-for-sale securities Investments in affiliates, accounted for using		1,501	188	1,313
the equity method		1,237		1,237
Property and equipment, net		1,247	39	1,208
Intangible assets not subject to amortization		13,675	5,292	8,383
Intangible assets subject to amortization, net Other assets, at cost, net of accumulated		2,492	908	1,584
amortization		98	19	79
Total assets	\$	24,676	7,089	17,857
Liabilities and Equity				
Current liabilities	\$	3,756	311	3,445
Long-term debt		6,406	300	6,106
Deferred income tax liabilities		2,844	853	1,991
Other liabilities		235	44	191
Total liabilities		13,241	1,508	11,733
Total stockholders' equity		6,936	1,208	5,728
Noncontrolling interests in equity of subsidiaries		4,499	4,373	126
Total equity		11,435	5,581	5,854
Total liabilities and equity	\$	24,676	7,089	17,587

Liberty Interactive Corporation Condensed Pro Forma Consolidated Balance Sheet March 31, 2014 (unaudited)

	Liberty Interactive historical (1)		Less: TripCo historical (2)	Liberty Interactive pro forma
<u>Assets</u>				
Cash and cash equivalents	\$	1,484	322	1,162
Other current assets		3,045	331	2,714
Investments in available-for-sale securities		1,470	284	1,186
Investments in affiliates, accounted for using the equity				
method		1,244		1,244
Property and equipment, net		1,272	73	1,199
Intangible assets not subject to amortization		13,683	5,294	8,389
Intangible assets subject to amortization, net		2,352	856	1,496
Other assets, at cost, net of accumulated amortization		112	21	91
Total assets	\$	24,662	7,181	17,481
Liabilities and Equity Current liabilities Long-term debt Deferred income tax liabilities Other liabilities Total liabilities	\$	3,586 6,608 2,831 271 13,296	383 290 836 48 1,557	3,203 6,318 1,995 223 11,739
Total stockholders' equity Noncontrolling interests in equity of subsidiaries		6,833 4,533	1,208 4,416	5,625 117
Total equity		11,366	5,624	5,742
Total liabilities and equity	\$	24,662	7,181	17,481

Liberty Interactive Corporation Condensed Pro Forma Consolidated Statement of Operations Year ended December 31, 2013 (unaudited)

	Liberty Interactive historical (1)	Less: TripCo <u>historical (2)</u>	Liberty Interactive pro forma
Revenue	11,252	1,034	10,218
Operating costs and expenses:	,	-,	,
Cost of sales	6,602	87	6,515
Operating and Selling, general and administrative, including stock-based			
compensation	2,587	649	1,938
Depreciation and amortization	943	315	628
	10,132	1,051	9,081
Operating income	1,120	(17)	1,137
Interest expense	(373)	(12)	(361)
Share of earnings (loss)	33		33
Realized and unrealized gains (losses) on	(22)		(44)
financial instruments, net	(22)		(22)
Other income (expense)	(48)	1	(49)
Earnings (loss) from continuing operations	710	(20)	520
before income taxes	710	(28)	738
Income tax (expense) benefit	(130)	55	(185)
Net earnings (loss)	580	27	553
Less net loss attributable to the	70	2.4	45
noncontrolling interests	79	34	45
Net earnings (loss) attributable to:	501	(7)	500
Liberty stockholders	501	(7)	508
Net earnings (loss) attributable to:	420	(25)	162
Liberty Interactive stockholders Liberty Ventures stockholders	438 63	(25) 18	463 45
ProForma basic net earnings (loss) attributable t	o Liberty stockholders per	common chara:	
Series A and Series B Liberty Interactive	\$ 0.8		0.89
Series A and Series B Liberty Ventures	\$ 0.8		0.61
Series A and Series B Liberty Ventures	\$ 0.8.	3	0.01
ProForma diluted net earnings (loss) attributable	e to Liberty stockholders po	er common share:	
Series A and Series B Liberty Interactive	\$ 0.83		0.88
Series A and Series B Liberty Ventures	\$ 0.8	5	0.61
Liberty Interactive commons stock			
Basic	519	9	519
Diluted	52	7	527
Liberty Ventures commons stock			
Basic	7-	4	74
Diluted	7-		74
Dilucu	/•	•	/4
	4		

Liberty Interactive Corporation Condensed Pro Forma Consolidated Statement of Operations Three months ended March 31, 2014 (unaudited)

	(unuuunteu)			
	Liberty Interactive historical (1)	Less: TripCo <u>historical (2)</u>	Liberty Interactive pro forma	
Revenue	2,728		294	2,434
Operating costs and expenses: Cost of sales	1,566		10	1,556
Operating and Selling, general and	1,500		10	1,330
administrative, including stock-based compensation	654		185	469
Depreciation and amortization	232		70	162
- ·F·······	2,452		265	2,187
Operating income	276		29	247
Interest expense	(99)		(2)	(97)
Share of earnings (loss)	(2)			(2)
Realized and unrealized gains (losses) on				
financial instruments, net	(25)			(25)
Other income (expense) Earnings (loss) from continuing operations	8	==		8
before income taxes	158		27	131
Income tax (expense) benefit	(48)		(4)	(44)
Net earnings (loss)	110		23	87
Less net loss attributable to the noncontrolling				
interests	28		18	10
Net earnings (loss) attributable to:				
Liberty stockholders	82		5	77
Net earnings (loss) attributable to:				
Liberty Interactive stockholders	110		(4)	114
Liberty Ventures stockholders	(28)		9	(37)
ProForma basic net earnings (loss) attributable to Series A and Series B Liberty Interactive Series A and Series B Liberty Ventures	\$ \$	0.22 (0.38)		0.23 (0.51)
ProForma diluted net earnings (loss) attributable				
Series A and Series B Liberty Interactive	\$ \$	0.22		0.23
Series A and Series B Liberty Ventures Liberty Interactive commons stock	2	(0.38)		(0.50)
Basic		494		494
Diluted		504		504
Liberty Ventures commons stock Basic		73		73
Diluted		74		74

Liberty Interactive Corporation Notes to Condensed Pro Forma Consolidated Financial Statements (unaudited)

- (1) Represents the historical financial position and results of operations of Liberty. Such amounts were derived from the historical consolidated financial statements of Liberty Media Corporation as filed on Form 10-K on February 28, 2014 and Form 10-Q on May 8, 2014, respectively, with the SFC
- (2) Represents the historical financial position and results of operations of Liberty. Such amounts were derived from the historical consolidated financial statements of Liberty TripAdvisor Holdings, Inc. as filed on Form S-1 as amended on August 8, 2014 with the SEC.